# Contracting and quality upgrading: evidence from an experiment in Senegal

Joshua W. Deutschmann<sup>1</sup>, Tanguy Bernard<sup>2</sup>, and Ouambi Yameogo<sup>3</sup>

<sup>1</sup>University of Chicago <sup>2</sup>University of Bordeaux <sup>3</sup>Islamic Development Bank

This version: January 19, 2023 Latest version: click here

#### Abstract

We conduct a randomized experiment with groundnut producers in Senegal to address barriers to quality upgrading and access to international markets. We offer a bundled flexible contract that provides credit, training, and a price premium guarantee to encourage use of a new quality-improving technology. Producers randomly offered the contract are significantly more likely to purchase the technology. In areas where quality is otherwise lower due to agro-climatic conditions, producers are significantly more likely to comply with international standards. Adoption falls two years after our intervention when a price premium guarantee is no longer available, but remains persistently higher among previously-treated producers. We find that producers change their commercialization behavior when the contract is active, opting to sell more to cooperatives and less to other buyers, but this change is small in magnitude and does not persist to the subsequent season.

Corresponding author: jdeutschmann@uchicago.edu. Yameogo was employed by the International Institute of Tropical Agriculture at the time of this study. The authors would first like to thank the farmers and management teams at our partner cooperatives for their investment and eager participation in the project. The authors additionally thank Laura Schechter for her tireless support and contributions to developing this line of research; Mariama Konte, Samba Mbaye, Medoune Sall, and the team at CRDES for their work implementing the project; Abiodun Opaleye, Lamine Senghor, and the IITA Aflasafe team for their support of the project; and Brad Barham, Jonathan Bauchet, Augustin Bergeron, Soumaïla Gansonré, Louise Guillouët, Molly Lipscomb, Ana Paula Melo, Aristide Ouédraogo, Alex Orenstein, Jacob Ricker-Gilbert, Safiétou Barro Samb, Adam Theising, Emilia Tjernström, and seminar and conference participants at AAEA, AFE, CSAE, DIL, GPRL, LEAP/LdA, MWIEDC, NBER, NEUDC, PEDL, Sherbrooke University, STEG, Toulouse School of Economics, USDA ERS, and the UW-Madison Agricultural and Applied Economics department for helpful comments and suggestions. This project received financial support from the CGIAR Research Program on Policies, Institutions, and Markets, the CGIAR Research Program on Agriculture for Nutrition and Health, and the Agricultural Technology Adoption Initiative. Deutschmann acknowledges additional support from the Development Innovation Lab and the UW AAE Kenneth and Pauline Parsons Graduate Fellowship Fund and SRGC Fund. This study received IRB approval from IFPRI (MTID-19-0628) and is registered with the AEA RCT Registry (AEARCTR-0006315). Any errors in this draft are the sole responsibility of the authors.

## 1 Introduction

Linking producers to global value chains can improve productivity, increase incomes, and improve welfare (Minten et al., 2009; Reardon et al., 2009; Atkin et al., 2017; Barrett et al., 2020; World Bank, 2020). To participate in these value chains producers must be able to meet international quality standards, which can present a major barrier to market access (Ferro et al., 2015; Fontagné et al., 2015; Bai, 2018; Fernandes et al., 2019). Meeting quality standards often requires producers to change practices or invest in new technologies. In low-income countries, existing evidence broadly supports the need to resolve liquidity and information constraints to encourage smallholder farmers to adopt new technologies. (Saenger et al., 2014; Atkin et al., 2017; Bernard et al., 2017; Abate et al., 2018; Macchiavello and Miquel-Florensa, 2019; Park et al., 2021). While often necessary, such support may not be sufficient when these technologies do not increase yields, as is often the case when it comes to production quality and food safety (Bold et al., 2022; Hoffmann et al., 2022). Bundled interventions that jointly facilitate the use of food safety technologies and guarantee market rewards for safe production may prove more effective than each one separately.

In this paper, we design and test an intervention that aims to increase producers' ability to meet quality standards sufficient to enter global value chains. We work with groundnut producers in Senegal who access inputs and credit through local cooperatives. While cooperatives do seek to support collective commercialization, sales through these groups are often limited to repayment of input credit. We test whether a new food safety technology can be promoted through cooperatives in two ways: simply providing information and access to the technology, or a bundled flexible contract that offers credit and a price premium guarantee. We present evidence that our bundled contract enables producers to invest in quality-improving technology, improves compliance with international food safety quality standards in high-risk areas, and impacts producer commercialization behavior.

The context of our study is groundnut cultivation in Senegal, where groundnuts are both the most valuable export crop and widely consumed locally. We focus on one specific food safety quality standard—aflatoxin contamination—which affects both export market access and public health. Aflatoxins are a Group 1 human carcinogen (IARC, 1993, 2012; National Toxicology Program, 2016) produced by a fungus (*aspergillus flavus*) which contaminates staple crops including groundnuts, maize and rice.<sup>1</sup> Many countries impose aflatoxin standards for imported commodities. Our data collection confirms that aflatoxin contamination is indeed a problem in the groundnut basin of Senegal: less than 70 percent of farmers in

<sup>&</sup>lt;sup>1</sup>One report called aflatoxins "amongst the most potent mutagenic and carcinogenic substances known" (EFSA, 2007). The carcinogenic risks of aflatoxin exposure are compounded by high rates of Hepatitis B in many low-income countries (Turner et al., 2003). Chronic exposure can also contribute to childhood malnutrition and immunosuppression (Gnonlonfin et al., 2013). Aflatoxin contamination is an increasingly salient concern for producers of a variety of staple crops. With climate change exacerbating changes in rainfall patterns (Clavel et al., 2013), aflatoxins may be responsible for up to one quarter of all liver cancer cases worldwide and more than four billion people may be chronically exposed (Williams et al., 2004; Liu and Wu, 2010; Liu et al., 2012).

the control group were in compliance with European Union (EU) import standards.<sup>2</sup> We focus on EU standards because they are the strictest in the world, and Europe is among the largest importing regions for groundnuts (Garcia-Alvarez-Coque et al., 2020; CBI, 2022).

Despite the health and economic consequences of aflatoxin contamination, at baseline we find low awareness of the problem among groundnut farmers. In partnership with two farming cooperatives in the "groundnut basin" of Senegal, we inform farmers in 40 villages about aflatoxins and a novel technology designed to fight them, the bio-control product Aflasafe<sup>®</sup>. This new technology allows farmers to treat their fields and prevent aflatoxin-causing *a. flavus* fungi from developing on crops. Before our project, Aflasafe was not commercially available in Senegal, so our project represents farmers' first exposure to the technology. The technology is relatively low-cost,<sup>3</sup> is simple to use, and may additionally provide lasting protection to crops during storage and transport.

We design a new bundled contracting intervention that offers credit, training, and a guaranteed price premium over the minimum price that cooperatives would otherwise offer. This intervention builds on the existing model of our cooperative partners, who typically sell inputs to farmers on credit, provide extension advice on their use, and provide a reliable outlet for farmers to sell their output at or above the government-set price floor.<sup>4</sup> We implement a cluster-randomized experiment to test this intervention with a sample of 396 farmers in 40 villages.

We find that the intervention is highly effective at increasing adoption of the food safety technology: adoption increases from about 10 percent in control villages to around 90 percent in treated villages. In other words, bundling input credit, information, and a price premium guarantee is sufficient to achieve nearly universal investment in quality upgrading for aflatoxin prevention in our context. This finding is somewhat unique in magnitude but is in line with a growing body of evidence on the potential of bundled interventions (Abate et al., 2018; Deutschmann et al., 2021; J-PAL and CEGA, 2022; Suri and Udry, 2022). Information and credit alone are often insufficient to achieve significant investments in quality, but pairing extension with a market for quality can increase investment and quality outcomes (Bold et al., 2022; Hoffmann et al., 2022).

On average, farmers in the treatment group were 12% more likely to comply with strict EU quality standards, but this effect is not statistically significant. We explore several candidate explanations for the difference between our results and previous agronomic evidence. One key factor, demonstrated by previous agronomic evidence, is that agroclimatic conditions

 $<sup>^{2}</sup>$ Although we focus on export market standards here, aflatoxin contamination is also highly relevant for local public health because most producers retain some production for home consumption. Previous work in Senegal found post-harvest aflatoxin contamination in groundnuts to be correlated with levels of aflatoxin-albumin adduct (AF-alb), a biomarker of aflatoxin exposure (Watson et al., 2015).

<sup>&</sup>lt;sup>3</sup>Treating one hectare of cropland costs about \$17 USD at market price. The mean production on one hectare (about 900 kgs) is worth about \$350 USD at 2018 market prices.

<sup>&</sup>lt;sup>4</sup>This arrangement mirrors resource-providing contracts that are common in many agricultural value chains (Minot and Sawyer, 2016; de Janvry and Sadoulet, 2020).

cause substantial differences in contamination risk across space (Cotty and Jaime-Garcia, 2007; Bowen and Hagan, 2015; Waliyar et al., 2015; Hendrix et al., 2019; Magnan et al., 2021). Using samples collected from control farmers and satellite data on growing season temperature and precipitation, we confirm the presence of spatially heterogeneous risk. We then predict which villages experienced the highest average risk of contamination given the conditions they faced. In areas where contamination risk was highest, the intervention has a large and statistically significant effect on quality standard compliance. In our preferred specification, treated farmers in high-risk areas produce groundnuts 49% more likely to comply with EU standards compared to control farmers in comparable areas.

Although we cannot fully disentangle the effect of each component of our bundled treatment, we provide some evidence that the availability of a price premium has a large impact on producer decisions to invest in quality upgrading. After the successful pilot through our project, our partner cooperatives decided to make the technology widely available to their members. They did so in the same fashion as they make other inputs available, providing training to village representatives and providing most members with credit for the purchase. Importantly, they did not provide any explicit guarantee of a price premium. Using administrative data from our partner cooperatives, we find that adoption of the food safety technology falls sharply two years after our intervention. About 11% of cooperative members chose to purchase the technology when it became widely available. Past experience with the technology increases follow-up adoption, with previously-treated participants 45% more likely to continue using the technology. By contrast, past exposure to information alone does not increase follow-up adoption, nor do we find evidence of spillovers within previously-treated villages.

Finally, we turn to farmer commercialization behavior and explore how our intervention impacts the choice of where farmers sell their output. Our study takes place in a context with market-level barriers to successful quality upgrading and contracting. Inconsistent but high spot market prices mean that many producers sell the bulk of their production to spot market buyers instead of through cooperatives, who are typically constrained to pay the government-set price floor.<sup>5</sup> Spot market buyers, especially those selling to the Chinese market, generally do not reward aflatoxin reduction.<sup>6</sup> Given the important role of the spot market, and the potential for significant disruption to that market as the Chinese government increases enforcement of food safety standards (Sun et al., 2021), it is important to consider how our intervention impacts the commercialization decisions of farmers. In the year of our study, spot market prices exceeded the typical price paid by cooperatives

<sup>&</sup>lt;sup>5</sup>Cooperatives often on-sell the bulk of the groundnuts they purchase to the parastatal groundnut oil company, who also serves as an important source of credit and inputs for cooperatives to provide to their members.

<sup>&</sup>lt;sup>6</sup>According to COMTRADE data on exports of unprocessed groundnuts, exports to China made up about 28 percent of total exports by weight on average in the first five years after Senegal lowered export barriers in 2010. In the subsequent five years from 2015 to 2019, exports to China grew to constitute 91 percent of total exports on average.

by 38% on average in nearby markets. In the year of our study, producers in control group opted to sell 48% of their groundnut production outside the cooperative and just 12% to the cooperative. This is comparable to the previous year (52% on the spot market, 17% to the cooperative), and mirrors findings from Aflagah et al. (2022) demonstrating the difficulty cooperatives face in aggregating a large quantity of output.

Our intervention takes a small step towards impacting the commercialization decisions of farmers. Treated farmers increased total output sales to the cooperative by about 65% relative to farmers in the control group. However, farmers in the treated group still elected to sell 41% of their output on the spot market which offers no aflatoxin-contingent pricing. Given high market prices outside the cooperatives, farmers with high-quality production faced a much smaller trade-off when choosing where to sell, whereas farmers with low-quality production were unambiguously better off selling on spot markets. Although our intervention improves output aggregation for the cooperative, the effect is relatively small in magnitude, about twice the value of the credit provided on average. We also find that changes in commercialization behavior do not persist to the subsequent season, when the contract was no longer available. Our paper provides a cautionary tale, in line with findings in Ashraf et al. (2009) and Bernard et al. (2017), that resolving individual-level constraints for smallholder farmers may be insufficient given market-level forces.

Putting these findings together, we interpret the success of the contract in increasing technology adoption as being akin to flexible price insurance against the volatility of spot market prices. If the spot market prices were to fall below the premium price, as they did in most of the previous ten seasons, the flexible contract would increase farmers' incomes. Furthermore, spot markets are likely to become more quality sensitive in the future, due to changes in demand and improvements in screening technology. One may additionally argue that the flexible nature of the contractual relationship between cooperatives and members, by which they are encouraged but not bound to sell their production to the cooperative, actually supports sustained adoption of food safety technologies. If side-selling does not prevent one from accessing the technology in future periods, it aligns incentives for farmers to sustainably engage with food safety technologies as long as there is a minimum price premium available. Cooperatives are also local member-owned institutions, which may explain their flexibility but also may support credit repayment above and beyond similar contracting arrangements offered by purely commercial actors.

The sustainability of food safety promotion through cooperatives still depends on their capacity to offer a price premium. This in turn depends on exporters' willingness and ability to purchase certified production from cooperatives, as well as the availability of financial resources to backstop a price premium guarantee. Our results suggest that contracts which require only a minimum quantity sufficient to repay input loans are effective at significantly changing producer behavior, and in years with low spot market prices would result in cooperatives aggregating a significant quantity of high-quality production. From a public health perspective, reducing human exposure to a dangerous toxin is a worthy objective in itself. Our findings suggest there may be a role for limited public price subsidies to backstop market prices and encourage widespread adoption. Of course, national governments may be more willing to pay for such subsidies for crops destined for local consumption. Combining export incentives with limited price support from governments may achieve widespread investment with limited reliance on public resources.

Our work contributes to the literature in three ways. First, we contribute to the substantial literature on agricultural technology adoption. Credit, information, and price uncertainty may all be relevant constraints limiting adoption of improved technologies in a particular context (Feder et al., 1985; Foster and Rosenzweig, 2010; Magruder, 2018; Suri and Udry, 2022). Credit in particular, by shifting the timing of payment or salience of the total cost, can increase adoption of a variety of technologies including fertilizer, bednets, and crop insurance (Duflo et al., 2011; Devoto et al., 2012; Tarozzi et al., 2014; Casaburi and Willis, 2018). A growing body of evidence suggests bundling interventions together to address several of these constraints can be highly effective at increasing technology adoption and farmer productivity (Abate et al., 2018; Deutschmann et al., 2021; Suri and Udry, 2022). We corroborate findings in the literature that a bundled intervention can impact technology adoption, in a novel context in which the technology addresses quality instead of yields.

Second, we contribute to a growing literature on contracting in settings with limited enforcement. Existing work suggests contract farming can be an effective avenue for farmers to improve productivity or quality (Arouna et al., 2021; Macchiavello and Miquel-Florensa, 2019).<sup>7</sup> Successful contracting typically depends on either effective institutions or informal relationships, often termed relational contracting (Brown et al., 2004; Michler and Wu, 2020). Previous work has additionally demonstrated the complementary nature of norms, incentives, and reciprocity in enforcing informal contracts (Charness and Haruvy, 2002; Fehr et al., 2009; Rigdon, 2009; Finan and Schechter, 2012; Kessler and Leider, 2012; Fahn, 2020). We contribute new field-experimental evidence that behavioral characteristics play a role in contracting and commercialization decisions.

Third, we contribute to a small but important literature on the determinants of aflatoxin mitigation by smallholder farmers. Given the public health implications of widespread aflatoxin exposure, especially in low-income countries, it is of direct policy interest to understand the best strategies for reducing contamination. Previous work on bio-control technology in particular has found that a price premium alone, or a price premium bundled with index insurance, can induce modest increases in technology adoption (Hoffmann et al., 2018b; Narayan et al., 2019; Hoffmann et al., 2022). More generally, information and price premiums can induce increased adoption of complementary aflatoxin mitigation strategies, like drying crops on a tarp or storing them in hermetic storage bags (Magnan et al.,

<sup>&</sup>lt;sup>7</sup>Contract farming can also affect other dimensions of farmer welfare, by increasing food security (Bellemare and Novak, 2017) or reducing income volatility (Bellemare et al., 2021).

2021; Bauchet et al., 2020). We demonstrate an intervention which is sufficient to achieve widespread adoption of bio-control technology.

The remainder of the paper is organized as follows. Section 2 provides additional context about aflatoxin and groundnut cultivation in Senegal. Section 3 presents the design of the experiment, and Section 4 provides an overview of the key covariates and outcomes of interest. Section 5 presents the empirical strategy and Section 6 presents the results of our study. Section 7 concludes.

## 2 Context

#### 2.1 Groundnuts in Senegal

Groundnut cultivation has represented a significant fraction of economic activity in Senegal since well before independence.<sup>8</sup> Groundnuts are grown by more than half of households in extreme poverty, use more than 40% of cultivated land, and are widely consumed in a variety of forms in Senegal (World Bank, 2015). In 2018, Senegal was the 7th largest producer of groundnuts in the world (USDA Foreign Agricultural Service, 2020). However, as quality standards and the nature of the global groundnut market shifted from oil and processed material to whole nuts, Senegal's share of the international groundnut trade fell from 17 percent in the early 1960s to a low of less than 1 percent in the 1990s. More recently, after easing export restrictions on whole nuts, Senegal has approached 10 percent of world trade in nuts, but with high levels of inter-annual volatility (World Bank, 2017).<sup>9</sup> Groundnut productivity largely stagnated after the 1960s and Senegal's share of world groundnut production has remained roughly stable at 2-3 percent since the 1990s (Kelly et al., 1996; World Bank, 2017).

The center of groundnut production is the city of Kaolack and the surrounding region, aptly termed the "groundnut basin." Shifting rainfall patterns in the groundnut basin since the 1980s have increased the risk of *a. flavus* development and aflatoxin contamination (Clavel et al., 2013). Consequently, dietary aflatoxin exposure in the groundnut basin is particularly high (Watson et al., 2015). Despite significant work to identify seed varieties more resistant to aflatoxin, agronomists have thus far failed to identify any variety which was completely resistant to aflatoxin contamination (Waliyar et al., 1994; Anderson et al., 1995; Holbrook et al., 2000; Clavel et al., 2013), although new work shows promise in developing aflatoxin-resistant seeds (Sharma et al., 2018).

<sup>&</sup>lt;sup>8</sup>Cultivation of groundnuts began in Senegal in the 1840s and grew quickly into a major export during the era of French colonial rule (Brooks, 1975).

<sup>&</sup>lt;sup>9</sup>Despite changes in the nature of the global market, the domestic market is still largely structured to protect groundnut oil producers (including a large state-owned company), with high implicit taxation eroding incentives for non-oilseed production (Masters, 2007; World Bank, 2017). However, in a sign of shifting priorities for rural producers and firms, repeated efforts to impose an export tax on whole nuts have largely failed under public pressure (Fofana et al., 2018).

Groundnut farmers in Senegal are typically members of a cooperative or rural-producers organization. Groundnut cooperatives in Senegal have roots in post-colonial political economy; the first president invested heavily in rural welfare through a robust state-controlled system of cooperatives (Casswell, 1984). In subsequent decades, state involvement in the cooperatives declined. Productivity also lagged as development of new seed varieties slowed and the parastatal groundnut oil producer faced a series of financial difficulties. In recent decades, many cooperative organizations have provided at best limited benefits to members (Bernard et al., 2008). However, work by the Senegalese government and international organizations has resulted in a new class of cooperative organizations more active in input provision and output commercialization (Clavel and Gaye, 2018; Eclosio, 2018). These cooperatives distribute seeds, fertilizer, and pesticides to farmers, typically on credit repayable in kind after harvest. Farmers express their input needs in the months before planting, and the cooperatives aggregate farmer requests to purchase inputs in bulk and re-sell them to farmers. These cooperatives also provide extension services to farmers, with trained technicians on staff and lead farmers active in many villages. They aim to commercialize output collectively, but in practice often pay only the government-set price floor and resell output to the quality-insensitive state-owned groundnut company.<sup>10</sup>

In this setting, farmer side-selling to buyers other than the cooperative is common, but inconsistent, as spot market prices (shown in Figure 1) can vary significantly from one year to the next. Cooperatives may punish complete non-compliance by restricting future access to credit or other services like plow rental. However, the main channel by which cooperatives encourage farmers to bring their production is positive in nature: namely, access to trials and new technologies. Baseline data in our sample suggests groundnut farmers in Senegal are generally willing to adopt technologies: more than 70 percent use fertilizer, 60 percent use pesticides or other products, 47 percent use improved seed varieties,<sup>11</sup> 98 percent use an animal-driven seeder, and 100 percent use an animal-driven plow. Many of the "new class" cooperatives in Senegal are active in developing and testing new seed varieties, new farming techniques, and new contracting arrangements (Clavel and Gaye, 2018; Eclosio, 2018). Conversations with cooperative leaders suggest they prefer to allocate access to these trials to farmers they see as reliable or highly skilled.

#### 2.2 Aflatoxins

The quality measure we study in this paper is aflatoxin contamination. Aflatoxins are toxic compounds produced by *aspergillus flavus*, a fungus which contaminates crops from soil and

<sup>&</sup>lt;sup>10</sup>The price floors that cooperatives and oil-press companies in Senegal follow are set annually by the National Inter-professional Groundnut Committee (CNIA), with some debate over how much the state influences the price chosen each year (Diagana, 2008). We show how these price floors compare to average annual market prices in Figure 1.

<sup>&</sup>lt;sup>11</sup>Many "improved" seed varieties widely used in the groundnut basin of Senegal are old varieties, introduced more than 30 years ago. However, 13 percent of farmers report participating in a seed trial at baseline.



Figure 1: Average annual market prices and government-set floor prices

Source: Commissariat à la Sécurité Alimentaire survey data and Comité National Interprofessionnel de l'Arachide reports. The market prices are averaged over the period November to February of each commercialization season and include all observations from the Kaolack and Fatick region.

spreads during storage (Frisvad et al., 2019). Aflatoxins affect a variety of staple and cash crops including maize, rice, and groundnuts (Udomkun et al., 2017). Across sub-Saharan Africa, research has found consistently high levels of human and animal aflatoxin exposure (Watson et al., 2015; Sirma et al., 2018; Blankson et al., 2019). Aflatoxin exposure has a variety of impacts on human and livestock health. Acute exposure to high levels of aflatoxins can be deadly (Probst et al., 2010; Kamala et al., 2018). Chronic exposure to lower levels of aflatoxins can cause child stunting, cancer, and immunosuppression (Coursaget et al., 1993; Wild, 2002; Hoffmann et al., 2018a; Voth-Gaeddert et al., 2018; Watson et al., 2018).<sup>12</sup> Aflatoxin exposure via contaminated feed products also affects livestock health and can transmit to humans via milk products (Bryden, 2012).

Governments and large buyers of crops across Africa have identified aflatoxin control as a key public health challenge (Partnership for Aflatoxin Control in Africa, 2015). The European Union began implementing harmonized aflatoxin standards in 2002 (Otsuki et al., 2001), with significant effects on Senegalese firms seeking to export whole groundnuts to European markets (Mbaye, 2005). This is not a problem unique to Senegal: existing work suggests firms and agricultural exporters are often constrained by stricter product standards in destination markets (Ferro et al., 2015; Fontagné et al., 2015; Fernandes et al., 2019).<sup>13</sup> African producers and exporters were particularly impacted by changes in EU standards (Agyekum and Jolly, 2017).

Aflatoxins are difficult to control because they are not directly observable: contaminated crops can look, smell, and taste identical to non-contaminated crops. Chemical tests for aflatoxins exist, but are not widely available, and the costs of consumable materials and testing equipment are non-trivial.<sup>14</sup> Early conversations with exporters and agro-processors during the design phase of this project suggest many are concerned about aflatoxins and eager to source low-aflatoxin groundnuts, but lacked the means to reliably identify them in the value chain.

Fragmented value chains between smallholder farmers and consumers, agro-processors and exporters make aflatoxin control along the chain a challenging and potentially costly proposition. Existing research on reducing aflatoxin incidence at the farmer level has found that low-cost practices (such as drying crops on a tarp) can reduce aflatoxin contamination (Turner et al., 2005; Magnan et al., 2021; Pretari et al., 2019; Bauchet et al., 2020; Jordan et al., 2020). However, the market rewards for a small farmer to reduce aflatoxin contamination are unclear. Consumer demand for reduced aflatoxin in local markets is inconsistent (Prieto et al., 2019; Hoffmann et al., 2020b,a). Contaminated crops are often sold to consumers in powdered or transformed form (Florkowski, 2014). Exporters and other quality-sensitive

 $<sup>^{12}</sup>$  One study hypothesizes that reducing a flatoxin exposure to non-detectable levels could reduce liver cancer cases in high-risk areas by 23% (Liu et al., 2012).

<sup>&</sup>lt;sup>13</sup>In addition, previous work in Senegal found that tightening standards for fruit and vegetable exports to the EU induced structural changes in the supply chain (Maertens and Swinnen, 2009).

<sup>&</sup>lt;sup>14</sup>In 2016, ICRISAT announced a new low-cost aflatoxin test kit that would be available for less than \$2 per test (compared to \$20-25 per test for existing kits). However, this technology is not yet widely available.

buyers typically do not work with small farmers directly.

#### 2.3 Bio-control for aflatoxin reduction

Agronomists have developed a new bio-control technology to fight against aflatoxin contamination.<sup>15</sup> Marketed under the umbrella brand name *Aflasafe*, this technology is designed to limit the development of toxic strains of *aspergillus flavus* in fields (Bandyopadhyay et al., 2019). In each country where Aflasafe has launched,<sup>16</sup> local strains of *a. flavus* are first collected to identify competitive strains which do not produce aflatoxins. These strains are isolated and replicated to produce products like Aflasafe SN-01, which launched in Senegal and the Gambia in 2019 after more than five years of efficacy trials (Senghor et al., 2020). Senegal is one of the first countries in Africa where Aflasafe is produced locally by a commercial partner.

The technology uses sterilized seeds (which will not grow) as a delivery mechanism, with the concentrated Aflasafe treatment applied as a seed coating. To protect a plot, farmers broadcast the Aflasafe-coated sterile seeds in the field 4-6 weeks after planting. The atoxigenic Aflasafe strains spread in the fields and prevent the aflatoxin-causing strains from developing on crops.

Compared to existing aflatoxin control strategies, Aflasafe has two key advantages. First, it may provide lasting aflatoxin protection even if storage conditions along the value chain are not always ideal. Agronomic research in Senegal has found that, even if non-treated and treated samples show similar aflatoxin levels immediately post-harvest, poor storage conditions will cause measurable differences in contamination in a matter of weeks.<sup>17</sup> Second, Aflasafe may be more cost effective than hermetic storage bags, another proposed solution for aflatoxin control. Treating one hectare of groundnuts with Aflasafe costs about \$17 USD at current market prices. By comparison, purchasing hermetic bags in Senegal to store the production from one hectare may cost \$40 or more (Bauchet et al., 2020).<sup>18</sup>

## 3 Research Design

We implemented this project in partnership with two cooperatives located in the groundnut basin of Senegal, in the Kaolack and Fatick regions (Figure B.1). These cooperatives are

<sup>&</sup>lt;sup>15</sup>While this technology is new for African contexts, similar aflatoxin bio-control products have been used commercially in the United States for more than 20 years in a variety of crops (Dorner and Lamb, 2006; Dorner, 2009; Doster et al., 2014).

<sup>&</sup>lt;sup>16</sup>As of September 2020, localized versions of Aflasafe are on sale in seven countries in Africa, with development at various stages in thirteen more.

<sup>&</sup>lt;sup>17</sup>Source: preliminary research results and correspondence from IITA agronomic field staff in Senegal. It is worth noting, however, that this result is not conclusive in the literature: at least one study found that bio-control technology alone is insufficient to offer lasting aflatoxin protection during storage (Kinyungu et al., 2019).

<sup>&</sup>lt;sup>18</sup>If farmers are able to re-use hermetic storage bags for several years, the cost of the two strategies becomes comparable.

active in providing services to members, including input distribution, access to credit, and agricultural extension support. They focus primarily or exclusively on groundnut production. In what follows, we refer to our partner cooperatives as Northern and Southern, indicating their location relative to the Saloum river. Each cooperative is organized into village sections, where each section typically has a president and one or more lead farmers.<sup>19</sup> Each cooperative has a membership of at least 1500 farmers divided into more than 50 village-level sections. Using the membership lists of each cooperative as a sampling frame, we selected a study sample of farmers from 40 villages. We initially sampled 10 farmers per village, with 5 replacement farmers available, and ended up with a final sample of 396 participating farmers after the baseline.

We assigned 20 villages each to treatment and control groups.<sup>20</sup> Village randomization was stratified at the rural commune level, the smallest level of administrative division at the rural level. All sampled farmers were offered a free aflatoxin test, and all farmers received the same information about Aflasafe. Farmers in the treatment villages were eligible to purchase Aflasafe with zero-interest credit, repayable in cash or in kind after harvest. They were also promised a minimum price premium, relative to the state-set price typically offered by cooperatives, of 40 CFA (about \$0.07) per kg conditional on the results of an aflatoxin test.<sup>21</sup> Farmers in control villages were eligible to purchase Aflasafe, but had to pay up front. They were also informed that they could have their production tested for aflatoxins, but with no promise of a price premium. Farmers in the treated group additionally received an ex-ante promise of assistance applying the product from trained cooperative extension agents.<sup>22</sup>

We conducted a baseline survey in June and July 2019, collecting detailed information about farming practices in the previous season and plans for the current season, as well as aflatoxin awareness, involvement with and trust in the cooperative, and reciprocity. We introduced Aflasafe to farmers at the end of the baseline survey. First, enumerators read a script explaining the health risks caused by aflatoxin exposure. Then, they explained how Aflasafe works, discussed with farmers how to use it, and showed farmers a video which demonstrated how to apply Aflasafe to a field.<sup>23</sup> Finally, they distributed to each farmer a ticket they could redeem with the cooperative to access 10kg of Aflasafe (to be paid cash or on credit, depending on treatment status), sufficient to treat one hectare of groundnuts.<sup>24</sup>

<sup>&</sup>lt;sup>19</sup>Lead farmers and village presidents are typically the channels by which the cooperatives diffuse information about practices and new technologies.

 $<sup>^{20}</sup>$ Because *aspergillus flavus* can spread between neighboring fields, we chose to randomize treatment at the village level rather than the individual level.

<sup>&</sup>lt;sup>21</sup>This is a gross premium, not inclusive of the amount farmers need to repay to cover the cost of their Aflasafe purchase. In the season we study, farmers who chose to repay their credit in kind would need to sell an additional 40-48 kgs to their cooperative.

<sup>&</sup>lt;sup>22</sup>After farmers made adoption decisions, both groups received a similar level of help applying the product. The difference between the two groups is the ex-ante promise of assistance.

<sup>&</sup>lt;sup>23</sup>The full script (translated to English) is shown in Appendix D. The video can be seen here: video link (Wolof).

<sup>&</sup>lt;sup>24</sup>Due to project budget and supply constraints, and to ease logistics, we standardized the bundle offer to

Farmers learned the details of their treatment assignment upon receiving this ticket, which included a unique code for each farmer as well as a reminder about the contract terms for treated farmers.

Next, in August 2019 we called all participants to inform them that Aflasafe had become available at their cooperative<sup>25</sup> and remind them of the terms of their treatment assignment. We additionally asked for the date they planted their groundnuts, and informed them of the suggested window for Aflasafe application based on their planting date (roughly six weeks after planting).

Aflasafe distribution was managed directly by the cooperatives, and was run similarly to how other inputs are distributed. This means that farmers in villages near to the cooperatives go directly to the warehouse to pick up inputs, whereas for villages further from the warehouse, the cooperative collects orders in advance and organizes a delivery by truck. Enumerators were present in the headquarters of each cooperative during distribution to ensure treatment status was respected (i.e., that control-group farmers who had to purchase the product up front did pay up front, and treatment-group farmers who could receive product on credit received the product on credit).

After the product was distributed, extension agents from each cooperative visited each village to help farmers apply the product correctly. Although the application process is simple—farmers broadcast 10kg of the product relatively uniformly over one hectare—cooperative agents wanted to ensure farmers were well-informed about the process. These trained agents helped ensure correct application and recorded the date on which farmers applied the product. All adopting farmers received the same extension support from the cooperative.

After harvest, in December 2019 and January 2020, we sampled and tested the groundnuts to determine aflatoxin levels for farmers in the sample. Some farmers delivered groundnuts for sale to the cooperative, and for these farmers we collected a sample from the groundnuts delivered for sale. However, market conditions meant buyers outside the cooperative were often paying farmers higher prices, so many farmers elected to sell their groundnuts elsewhere and simply reimburse the cooperative in cash to cover their credit. For these farmers, we worked with village section presidents and agents from the cooperative to collect samples. Despite offering farmers a significant premium for a sample sufficient to measure aflatoxin levels, we were ultimately only able to test 83% of farmers. Some attrition at this stage was due to crop failure, whereas others reported selling their entire crop before we tried to collect a sample. To test the level of aflatoxin, we conducted lateral flow tests using a Neogen Raptor reader and standard sampling and testing procedures used by IITA for all

farmers as a single package of 10kg. Given the highly limited supply of Aflasafe in Senegal in the year of our study, farmers could not have sourced an additional quantity from other sources.

<sup>&</sup>lt;sup>25</sup>When we initially surveyed farmers during the baseline survey, we expected Aflasafe to be available for them no later than the end of July. Due to delays in production by the local manufacturer, the product was not actually available until late August.

Aflasafe development activities.

Farmers from treated and control villages who chose to adopt Aflasafe and achieved aflatoxin levels less than 4 parts per billion (ppb) received a premium price from the cooperative for their certified groundnuts.<sup>26</sup> This premium was paid only for bags delivered to the cooperative in advance of testing, and the resulting bags were sampled and certified following the test. Farmers were paid 250-275 CFA (\$0.43-0.47) per kg for their certified production–the final price was set by each cooperative depending on their logistical overhead costs. This represented a significant premium over the government-set price of 210 CFA (\$0.36) per kg, although anecdotal reports and market survey data suggest high export demand from Chinese buyers resulted in comparable or higher prices in local markets irrespective of aflatoxin levels.

Finally, we conducted an endline survey with farmers in June 2020. This endline survey was conducted by phone due to the COVID-19 crisis. We collected aggregate farmer-level estimates of groundnut production, output sales, and revenue. We also elicited endline awareness of aflatoxin and recall of their aflatoxin test results.

## 4 Data

In this section we describe the characteristics of the study population and the market environment. We additionally present information and summary statistics about the outcomes of interest.

#### 4.1 Baseline

Table 1 presents summary statistics and baseline balance tests. The characteristics of our sample motivate the importance of our project. In particular, note that the median farmer does not have a savings account (either in a bank or with a mobile money provider), consumes some of his output, and was unaware of aflatoxin at baseline. Most farmers have experience adopting fertilizer and pesticides, but yields per hectare are relatively low by global standards at about 900 kgs/hectare.<sup>27</sup>

We elicited several measures of potential behavioral mechanisms, following our Pre-Analysis Plan, using questions drawn from Falk et al. (2016, 2018). In particular, we elicited measures of intrinsic reciprocity, patience, and risk aversion (see Appendix D.2 for the exact question wording we used). Intrinsic reciprocity plays an important role in repeated interactions, and individuals may reward past kindness or punish past unkindness (Sobel,

<sup>&</sup>lt;sup>26</sup>For control farmers, this was unexpected, as they received no guarantee of a price premium at baseline. If the premium available from reselling these low-aflatoxin groundnuts was less than the premium promised to treatment group farmers, control farmers would have received a lower price.

<sup>&</sup>lt;sup>27</sup>Groundnut yields in China (the largest producer in the world) recently exceeded 3.5 metric tons per hectare. By comparison, yields in India (the second largest producer) are more comparable at 1-1.3 metric tons per hectare (USDA Foreign Agricultural Service, 2020).

	(1)	( <b>0</b> )	
	(1)	(2)	D:C
<b>T</b> 7 • 11	Control	Treatment	Difference
Variable	Mean (SD)	Mean (SD)	(2)-(1) [p-value]
Demographic variables			
Married $(0/1)$	$0.92 \ (0.27)$	$0.91 \ (0.29)$	-0.01 [0.62]
Polygamous marriage $(0/1)$	0.38(0.49)	$0.46 \ (0.50)$	$0.09 \ [0.14]$
Female $(0/1)$	0.36(0.48)	$0.31 \ (0.46)$	-0.04 [0.52]
Household head $(0/1)$	0.63(0.48)	0.62(0.49)	-0.01 [0.82]
Completed secondary school [resp.] $(0/1)$	$0.11 \ (0.31)$	0.10(0.30)	$0.00 \ [0.91]$
Completed sec. school [any in HH] $(0/1)$	$0.57 \ (0.50)$	0.60(0.49)	0.02  [0.79]
Household size	16.17(10.11)	16.39(9.33)	$0.13 \ [0.89]$
Children in household	6.63(4.77)	6.99(5.11)	0.31  [0.56]
Age	49.11(13.01)	47.66(12.82)	-1.48 [0.32]
Agricultural variables			
Aware of aflatoxin $(0/1)$	$0.09 \ (0.29)$	0.13(0.34)	$0.04 \ [0.25]$
Savings account $(0/1)$	0.34(0.47)	$0.31 \ (0.46)$	-0.03 $[0.57]$
Lead farmer $(0/1)$	$0.11 \ (0.31)$	$0.15 \ (0.35)$	0.03  [0.45]
Used fertilizer $(0/1)$	0.66(0.47)	0.75~(0.43)	0.08  [0.17]
Used pesticides $(0/1)$	$0.62 \ (0.49)$	$0.58\ (0.50)$	-0.04 [0.51]
Consumes some output $(0/1)$	$0.72 \ (0.45)$	0.74(0.44)	-0.01 [0.90]
Sold to cooperative $(0/1)$	$0.29 \ (0.46)$	$0.40 \ (0.49)$	$0.10 \ [0.14]$
Sold to other traders $(0/1)$	$0.77 \ (0.42)$	0.75~(0.43)	-0.01 [0.87]
Kept as seeds or given away $(0/1)$	$0.71 \ (0.46)$	$0.71 \ (0.46)$	0.01  [0.78]
Groundnut hectares cultivated	3.56(2.64)	3.40(2.52)	-0.09 $[0.74]$
Groundnut yield (kgs/hectare)	$859.31 \ (678.48)$	$909.49 \ (922.68)$	36.87  [0.59]
Recent cooperative member	$0.30 \ (0.46)$	$0.24 \ (0.43)$	-0.06 $[0.55]$
Behavioral variables			
Risk loving $(0/1)$	$0.21 \ (0.41)$	0.24(0.43)	0.03  [0.34]
Patient $(0/1)$	0.79(0.41)	0.78(0.41)	-0.01 $[0.78]$
Reciprocal $(0/1)$	$0.45 \ (0.21)$	$0.49 \ (0.25)$	$0.04 \ [0.06]$
p-value, $F$ -test of joint orthogonality across	s groups (asympt	otic)	0.00
p-value, $F$ -test of joint orthogonality across	s groups (empiric	al CDF)	0.55
Number of observations			396

Table 1: Baseline balance and summary statistics

Note: standard errors for differences for each baseline variable are clustered at the treatment assignment (village) level. Individual balance tests include commune fixed effects to account for randomization stratified at commune level. The *p*-value for the asymptotic test that observations are jointly orthogonal across groups is estimated using OLS, with treatment assignment as the dependent variable, all baseline covariates as independent variables, commune fixed effects, and standard errors clustered at the treatment assignment level. The *p*-value for the empirical CDF test is estimated using 1000 placebo draws that re-assign treatment at the village level, within commune strata, and computing the share of placebo *F*-statistics larger than the actual test statistic (Hansen and Bowers, 2008).

2005; Cabral et al., 2014). We hypothesized that highly reciprocal individuals may be more affected by a contract offer; namely, high-reciprocity treated farmers may be more likely to adopt the technology and sell output to the cooperative. We additionally hypothesized that

patience and risk aversion would moderate adoption decisions: for control farmers, adoption required payment now for a possible benefit in the future, whereas treated farmers could delay payment until harvest and had more certainty about the potential benefits.

Table 1 additionally presents balance tests. We test for balance individually and jointly across treatment and control groups. We find only one variable (reciprocity) with a statistically significant difference at the 10% level. To test joint balance, we first implement the conventional asymptotic test, regressing the treatment dummy on all the variables presented in Table 1, with commune dummies included and standard errors clustered at the village level. Despite failing to find a significant difference in any individual variable, this test does reject that treatment is jointly orthogonal to all baseline variables. However, as Hansen and Bowers (2008) point out, when the number of covariates is "large" relative to the number of clusters, asymptotic tests may over-reject the null. Therefore we additionally conduct a randomization inference procedure (Heß, 2017), taking placebo draws of treatment status at the village level (stratified by commune), and repeating the regression of placebo treatment status on the set of baseline covariates to generate an empirical CDF of F-statistics. We fail to reject the null of joint orthogonality using this approach. There is some disagreement in the literature about how to account for balance, or imbalance, in a randomized trial (Imai et al., 2008; Bruhn and McKenzie, 2009; Mutz et al., 2019; Snyder and Zhuo, 2020). In our preferred specifications below, we control only for commune, the level at which cluster randomization was stratified. However, we also present results which control for all baseline covariates shown in Table 1, which rarely leads to any change in the sign, magnitude, or statistical significance of our results.

#### 4.2 Adoption and Intentions

Next, we turn to our primary outcome measure: adoption of the new food safety technology. We observe two potential measures of adoption, based on administrative data and self-reported endline data. The administrative measure relies on two datasets shared by our partner cooperatives: the administrative logs from distribution, and field visit logs by extension agents. These two datasets coincide for 94% of observations.<sup>28</sup> In the analysis that follows, we use a harmonized measure which flags a farmer as having adopted if either the distribution logs or field visit logs indicate adoption. However, results are robust to using a variable defined by each underlying log file individually.<sup>29</sup> The second measure relies instead on the self-reported use of Aflasafe from the endline survey. As Figure B.3 demonstrates, self-reported Aflasafe use is slightly higher for control farmers and slightly lower for treated

 $<sup>^{28}</sup>$ For the other 6%, about 2% are flagged as purchasing Aflasafe without applying it, and about 4% are flagged as receiving extension assistance applying Aflasafe without purchasing it.

<sup>&</sup>lt;sup>29</sup>We additionally sent enumerators to audit a randomly-selected 50% of villages and confirm the technology was distributed and applied. In each village, they spoke to up to two randomly-selected adopters and non-adopters (as defined by the field visit logs). In only one case did they find a respondent flagged as a non-adopter in field visit logs but who reported receiving and applying the technology. In one case they also found a respondent flagged as an adopter who received but elected not to apply the technology.

farmers compared to the administrative measure. The difference in these two measures could indicate some leakage from treated farmers to control farmers, even though treatment was randomized at village level and adopting farmers received assistance and supervision applying the product.<sup>30</sup>

#### 4.3 Quality

The quality measure we study is aflatoxin contamination. As described above in Section 3, we collected samples from farmers in the first six weeks of the commercialization season and tested them for aflatoxin. Our primary quality-related outcome of interest is whether or not farmers produced groundnuts in compliance with the strict European Union standards for aflatoxin contamination. Figure 2 shows that only 66% of control group farmers were in compliance with those standards, suggesting high incidence of contamination. This result is in line with results from agronomic trials conducted elsewhere in Senegal in the same year (see Figure B.2).

We focus on a binary measure of quality standard compliance for two reasons. First, this is the most salient threshold for exporters, and therefore of particular importance for farmers and intermediaries seeking to access the lucrative European market. Second, our testing procedure has a minimum level of detection of 2 parts per billion (well below the EU quality standard). That is, any results below 2 only tell us that the sample was not contaminated, but do not tell us the exact level of aflatoxin detected. This does not affect our analysis of the EU cutoff, but does affect our analysis of the continuous outcome. Importantly, 160 out of 328 samples tested fall below 2 ppb, so this potentially impacts a large portion of our sample. We set any test results equal to the midpoint (1 ppb) if the recorded result was less than or equal to 2 ppb.

In the absence of Aflasafe use, aflatoxin levels can vary significantly across space and time. Figure 3 shows the commune-level average rates of contamination among control farmers in our sample. The figure demonstrates that average rates of EU phytosanitary standard non-compliance range from below 16 percent in one commune to more than 46 percent in others. This is not out of line with previous findings in the literature. Magnan et al. (2021) observe even higher spatial variation in contamination among groundnut samples in Ghana over three seasons. Waliyar et al. (2015) find 59-66 percent of groundnut samples with contamination above 4 parts per billion in different districts in Mali across two seasons.

An additional issue with measuring the effects of the contract on quality is sample attrition. As mentioned above, for farmers who delivered output to the cooperative, we sampled that output to test for aflatoxin levels. However, with many farmers electing to sell no output to the cooperative, we had to adapt our data collection strategy to collect

 $<sup>^{30}</sup>$ To rule out spatial spillovers in adoption, we estimate a regression specification similar to the one used in Miguel and Kremer (2004), in which we include the average treatment status or adoption decision in nearby villages. After controlling for a village's treatment status, these additional variables have no statistically significant effect on self-reported or admin data adoption.



Figure 2: Distribution of aflatoxin levels among farmers in the control group

Note: This figure shows the result of aflatoxin tests done on samples collected from farmers in the control group. The first line, marked "EU", corresponds to the legal limit on aflatoxins imposed by the European Union (4 ppb). The second line, marked "US/China", corresponds to the legal limit on aflatoxins imposed by the US and China (20 ppb).

samples from these farmers. We offered a significant premium to purchase a 1 kg sample from all farmers, not only those who delivered output for sale to the cooperative. In the end, we collected samples from 83% of participating farmers. Some farmers who did not deliver a sample experienced very low yields or total crop failure due to a challenging rainy season. Based on endline survey data, four percent of farmers had zero output, and a further eleven percent of farmers harvested only enough to cover the median baseline quantity kept for seeds and home consumption. Just under half of our attrition falls in this category. Other farmers, facing unusually high spot market prices at the start of the commercialization season, reportedly quickly sold their entire output before we re-contacted them to request a sample. Treated farmers were significantly less likely to deliver a sample for testing (11 percentage points) than control farmers. It is possible that treated farmers with particularly high quality were more easily able to sell their output for a high price on the spot market. Conversely, it is possible that treated farmers with particularly low quality did not wish to undergo quality certification for reputational reasons. Below, in Section 6.2.1, we discuss a variety of techniques to assess the robustness of our results given this attrition.



Figure 3: Incidence of aflatoxin contamination in control villages varies by commune

Note: This figure shows commune-level average rates of non-compliance with EU phytosanitary standards for aflatoxin contamination (4 parts per billion). The sample is restricted to control group farmers. The color scale groups communes by quartile of contamination rates.

#### 4.4 Commercialization Behavior

Finally, we consider farmers' commercialization decisions. As shown above in Table 1, at baseline farmers typically sold some output via the cooperatives and some output via other traders. Figure 4 shows the fraction of farmers that sold their output to the cooperative and to other buyers, as well as keeping some output for consumption and as seeds. This suggests many farmers who are ostensibly active cooperative members are not selling any output to the cooperative, and instead selling to other local buyers. Figure 5 shows how the mean farmer allocated their output across sales, consumption, and other uses at baseline.

These figures suggest several notable features of our setting. First, as in Aflagah et al. (2022), we find that farmers allocate a large fraction of their output to commercial sale outside of their cooperatives, and only a small fraction to the cooperatives. Second, there is a small but important fraction of output that is kept for home consumption. This is particularly relevant given that consumption of aflatoxin-contaminated crops can affect health of adults and children. Similarly, farmers keep a fraction for seeds or giveaways.<sup>31</sup>

<sup>&</sup>lt;sup>31</sup>Groundnuts are often given to others as payment-in-kind for labor during the growing season.

The quantity in levels kept for seeds is relatively constant across baseline and endline, despite lower yields in the endline season, suggesting farmers may aim to keep a minimum threshold to ensure future production regardless of yields or prices.



Figure 4: Fraction of farmers who sold, consumed, and kept any output at baseline

Figure 5: Mean fraction of output sold and kept at baseline



We analyze two outcomes relevant to understanding farmers' commercialization decisions. Figure 4 summarizes dummy variables equal to one if a farmer sold any output to the cooperative or another buyer, kept any output for home consumption or seeds, or gave any output away to others. We define corresponding variables using our endline data. The second outcome is the quantity of output allocated to the cooperative. Figure B.5 shows the distribution of output allocation to the cooperative at baseline and endline. The leftmost bar shows farmers who allocated zero output to the cooperative. The distribution is highly skewed at both baseline and endline. Given these two facts, we analyze the inverse hyperbolic sine transformation of output allocated to the cooperative.

We also face some attrition in these outcomes. The endline survey was conducted by phone in June 2020, and some farmers were unreachable after repeated attempts. We successfully surveyed 93% of the baseline sample, and attrition from the endline is not different by treatment status. Of the 7% who did not complete the endline survey, most were simply due to difficulty contacting the respondent by phone.

#### 4.5 Climate data

To estimate the relationship between aflatoxin and growing season agro-climatic conditions, we use two sources of remotely-sensed data on precipitation and land surface temperature. For precipitation data, we rely on CHIRPS 2.0 daily rainfall estimates. For land surface temperature, we rely on the Copernicus LST10-DC dekadal land surface estimates. Both datasets have a 0.05° spatial resolution. We consider a "growing season" period defined by planting dates reported by farmers and the average time to harvest for the groundnut varieties common in our sample (90-110 days).

We define two village-level growing season variables: number of dry spells and average growing season daily max temperature. To define dry spells, we follow Bowen and Hagan (2015) who find that 3-day and 4-day dry spells are predictive of aflatoxin contamination. We calculate the cumulative number of dry spells experienced in each village during the growing season period. To define average growing season daily max temperature, we again follow Bowen and Hagan (2015) and take the max temperature for each dekadal temperature estimate, and then take the mean of these values over the growing season. To match villages to the spatial datasets, we take the mean value of all pixels within a 1 kilometer radius of the village midpoint.<sup>32</sup>

## 5 Empirical Strategy

For our outcomes of interest we estimate the following equation via OLS:

$$Y_{ijk} = \alpha + \beta_1 T_{jk} + \delta \mathbf{X}_{ijk} + \gamma_k + \epsilon_{ijk} \tag{1}$$

 $<sup>^{32}</sup>$ We use the village midpoint because we only observe plot locations for 27 farmers in our dataset who were randomly selected for inclusion in the "audit" survey described above. The median distance from the plot to the village centerpoint is 0.97 km, motivating our use of a 1km radius. All results presented are robust to using a larger radius.

where  $T_{jk}$  is the treatment assignment of farmer *i* in village *j* in commune *k*,  $\gamma_k$  is a commune fixed effect, and standard errors  $\epsilon_{ijk}$  are clustered at the treatment assignment (village) level. We present results with and without baseline controls  $\mathbf{X}_{ijk}$ . Where appropriate, we also estimate the treatment effect on the treated via 2SLS where we instrument Aflasafe adoption by treatment status.<sup>33</sup>

We additionally estimate the following equation:

$$Y_{ijk} = \alpha + \beta_1 T_{jk} + \beta_2 (T_{jk} \times H_{ijk}) + \beta_3 H_{ijk} + \delta \mathbf{X}_{ijk} + \gamma_k + \epsilon_{ijk}$$
(2)

where  $H_{ijk}$  is a measure of heterogeneity. We consider spatial heterogeneity (by predicted aflatoxin risk), as well as heterogeneity across behavioral measures presented above in Table 1: risk aversion, patience, and reciprocity. We again cluster standard errors at the village level, and additionally bootstrap standard errors when we include the predicted aflatoxin risk dummy.

We prepared a pre-analysis plan (PAP) in the course of developing this project, which is registered with the AEA registry (AEARCTR-0006315).<sup>34</sup> We deviate from the PAP in three ways, which we describe here briefly and more fully below in Appendix A.1. First, we implemented this project with a single treatment group, without offering any "partial" contracts as originally planned. Second, we randomized treatment assignment at the village level, to facilitate implementation and minimize potential spillovers. Third, we pre-specified several behavioral hypotheses which are infeasible to test due to insufficient variation in our elicited measurements.

Additionally, to address possibly non-random attrition in our measure of quality, we implement multiple strategies to test the robustness of our results. First, we implement Horowitz and Manski (2000) bounds which require no assumptions about the distribution of the missing data. Second, we implement a multiple-imputation procedure in which we predict the value of the missing outcomes using baseline farmer characteristics  $\mathbf{X}_{ijk}$ .

### 6 Results

In this section, we consider four main families of outcomes. First, we focus on farmers' investment in quality upgrading., using both administrative and self-reported measures. Second, we estimate intent-to-treat (ITT) and treatment-effect-on-the-treated (TOT) effects of our intervention on quality outcomes (aflatoxin standard compliance). Third, we explore dynamics in quality investment, looking at investment in the quality-improving technology

<sup>&</sup>lt;sup>33</sup>Exclusion restriction: we assume that that treatment only affects outcomes through the uptake of Aflasafe, since Aflasafe was not available in previous years and other inputs were available and priced similarly in both groups.

<sup>&</sup>lt;sup>34</sup>The PAP was presented publicly at the Northwestern University GPRL Pre-Analysis Plan Mini-Conference in May 2019, before we began any project activities in the field. This un-modified document was only submitted to the AEA registry in August 2020.

two years after our intervention. Finally, we explore farmers' commercialization decisions.

#### 6.1 Investment in quality upgrading

We first present results on investment in quality upgrading, i.e., farmers decisions to adopt the Aflasafe technology. Table 2 demonstrates the treatment had a remarkably large effect on farmers' adoption decisions using either measure of adoption. The treatment effect when we use the admin data is 79-80 percentage points, whereas the treatment effect using the self-reported outcome is 59-61 percentage points. These effects are robust to the inclusion of commune or cooperative FE and additional baseline controls described above in Table 1.

It is worth taking a moment to discuss the magnitude of these results. Because the intervention mirrors a resource-providing contract, our adoption measure nests credit uptake by treated farmers. Existing work on credit expansion typically finds low rates of credit adoption, in the range of 17-31% (Angelucci et al., 2015; Crépon et al., 2015; Tarozzi et al., 2015; Chowdhury et al., 2020). Similarly, existing work on credit expansion typically finds small impacts on technology adoption and input use (Crépon et al., 2015; Tarozzi et al., 2015; Beaman et al., 2020) or even no effect at all (Chowdhury et al., 2020; Nakano and Magezi, 2020).<sup>35</sup>

The bundled treatment additionally provided farmers ex-ante certainty about receiving training on proper use of the technology.<sup>36</sup> This is somewhat distinct from information interventions in the literature, which typically randomize the provision of information. Magnan et al. (2021) find information provision increases purchases of drying sheets for aflatoxin reduction by 9.7-14 percentage points. Training and farmer field days have been found to increase adoption of pest control practices and improved seeds by 12-15 percentage points (Emerick and Dar, 2020; Lerva, 2020). Information on international quality standards can similarly increase adoption of standards-compliant management practices in dragonfruit production (Park et al., 2021).

Finally, the treatment provided farmers with increased price premium certainty upon adoption and proper use of the technology. Results in the literature on the impact of a quality price premium are mixed. Hoffmann et al. (2022) find that a modest market premium for low-aflatoxin maize increases Aflasafe adoption by 75%, and Bold et al. (2022) find that offering a reliable market for quality maize increased adoption of post-harvest quality upgrading practices (proper drying, sorting, and winnowing) by 68-107%. By contrast, Magnan et al. (2021) find no significant effect of a low-aflatoxin price premium on the purchase of a low-cost technology (drying sheets) for groundnut farmers. Results on the impact of price certainty in general suggest a relatively small impact on farmer investment. Arouna et al. (2021) find that a contract with only price certainty is insufficient to increase

<sup>&</sup>lt;sup>35</sup>Along the same lines, relaxing credit and risk constraints via grants and index insurance has a significant but relatively small effect on input investment (Karlan et al., 2014; Bulte et al., 2019).

<sup>&</sup>lt;sup>36</sup>All farmers received in-person assistance to apply the technology. The difference between treatment and control is in the ex-ante promise of a field visit.

agricultural investment, although it can increase productivity. Karlan et al. (2011) test the impacts of crop-price indemnification embedded in agricultural lending, and find modest impacts on high-risk agricultural investment and on the probability of sale to higher-return buyers.

Our setting differs in several important ways from these past studies. First, our treatment offers relatively small loans which are exclusively intended to finance adoption of the new technology. Second, the technology is not expected to increase yields.<sup>37</sup> Instead, by increasing quality, farmers may expect to earn a higher price for their output. The intervention offered treated farmers increased certainty that adoption would be profitable, conditional on quality certification. Profitability is an important element of agricultural technology adoption decisions (Michler et al., 2019). Third, farmers face non-pecuniary incentives to adopt the technology, since it can also have health impacts for farmers who consume some of the groundnuts they grow. Fourth, there may be important complementarities between the components of our bundled treatment which increase adoption rates beyond the additive effects we might expect from each component.

	Admir	n Data Ad	loption	Self-Reported Adoption			
	(1)	(2)	(3)	(4)	(5)	(6)	
Treated	0.79***	0.79***	$0.78^{***}$	0.59***	$0.61^{***}$	$0.61^{***}$	
	(0.06)	(0.06)	(0.06)	(0.07)	(0.06)	(0.06)	
Observations	396	396	396	370	370	370	
$R^2$	0.621	0.647	0.673	0.347	0.377	0.425	
Control Mean Dep. Var	0.10	0.10	0.10	0.20	0.20	0.20	
Commune FE	Ν	Υ	Υ	Ν	Υ	Υ	
Baseline controls	Ν	Ν	Υ	Ν	Ν	Υ	

Table 2: Aflasafe adoption

Results in this table are from linear regressions of the adoption dummy on the treatment dummy. Admin Data Adoption is measured using distribution logs and extension agent field visit logs, provided by our partner cooperatives. Self-Reported Adoption was elicited in the endline survey. Standard errors (in parentheses) are clustered at the treatment assignment (village cluster) level. Baseline controls included are all variables shown above in Table 1.

While we cannot fully address the mechanisms driving the treatment effect on adoption, we collected additional data which is suggestive of the importance of credit constraints and time consistency.<sup>38</sup> At baseline, after introducing farmers to Aflasafe and providing the conditions of their treatment assignment, we collected a non-binding measure of intention to purchase the product. Additionally, in the week before distribution began, we called farmers to inform them the product would soon be available and collect a second non-binding

<sup>&</sup>lt;sup>37</sup>In our data, there is no detectable effect of technology adoption on yields.

 $<sup>^{38}</sup>$ We did attempt to measure time inconsistency directly in our baseline survey, but we observe very little variation in the resulting measurement, with only 10% showing any evidence of time inconsistency in a non-incentivized hypothetical exercise.

intention to adopt. Table B.2 demonstrates statistically significant treatment effects on these non-binding measures that are much smaller in magnitude than those we observe in the ultimate adoption decision. Control farmers were highly interested in the product at baseline and optimistic about their intent to adopt, with more than 90% indicating interest. At the time when distribution began, and farmers in the control group would need to pay cash to purchase the product, a majority of control group farmers remained optimistic about their ability and intention to purchase the product. The contract treatment seems to have allowed farmers to follow through with their intentions, rather than fundamentally shifting farmer interest in the technology.<sup>39</sup>

We may also learn more about mechanisms driving our effects by exploring heterogeneity in adoption. Ex ante, we anticipated heterogeneity in adoption decisions by farmer behavioral characteristics like reciprocity, patience, and risk aversion. We test for heterogeneity along these dimensions and report results in Table B.1. Our results have consistent sign across selfreported and admin data adoption measures, although we observe no statistically significant heterogeneity using the self-reported measure. Given the large average treatment effect and limited variation in some of our behavioral measures, it is perhaps unsurprising that we find only limited evidence of heterogeneity in this outcome along any dimension. Nevertheless, the sign of the interaction term between treatment and risk preferences in particular is consistent with our pre-specified hypothesis about the role of the contract (and the premium guarantee in particular) in affecting adoption decisions.

#### 6.2 Evidence on quality improvements and Aflasafe effectiveness

Next, we present results on quality outcomes (aflatoxin contamination). Our study is among the first to explore the results of Aflasafe use under real world conditions which may deviate from tightly-controlled agronomic trials. We consider two outcomes: EU phytosanitary standard compliance<sup>40</sup> and a continuous measure of aflatoxin contamination.

Table 3 demonstrates the impact of the contract on phytosanitary compliance. We see that on average, the point estimate on standard compliance is positive but not statistically significant. This may seem a surprising departure from results in controlled agronomic studies (Bandyopadhyay et al., 2019; Senghor et al., 2020). Given that about 90% of farmers in the treated group purchased Aflasafe, agronomic results would suggest we should see no more than 10% of samples from that group showing any contamination levels. In practice, we observe that about 27% of samples from treated farmers exceeded EU phytosanitary limits.

<sup>&</sup>lt;sup>39</sup>This interest in the product persists after our experiment. When contacted by phone after the conclusion of the experiment, 92% of farmers expressed an interest in using Aflasafe in the next season, without any information about credit access or other conditions of a potential contract. There is no statistically significant difference in this measure of intended adoption across treatment and control groups.

 $<sup>^{40}</sup>$ These are the strictest aflatoxin quality standards in the world, requiring total aflatoxin levels less than 4 p.p.b.

Several factors may help rationalize our results with the agronomic evidence. First, the samples we tested may not have come from farmers' treated fields. To simplify implementation and due to budget and Aflasafe availability constraints, our intervention offered farmers 10 kgs of Aflasafe, sufficient to treat one hectare. On average, farmers in our sample planted about 3.5 hectares, so the majority of the output in the treated group would not have been treated with Aflasafe. Second, due to production delays the distribution of Aflasafe happened later than originally planned. As a result, just 9% of farmers received and applied the product within the recommended 4-6 week window after planting. The median farmer applied the product nearly 9 weeks after planting. This likely dampened the possible effectiveness of the product.

However, a key factor that emerges from the agronomic literature on aflatoxins is spatial heterogeneity in contamination risk. For example, recent work by Magnan et al. (2021) in Ghana finds substantial variation in aflatoxin levels: in their baseline season, contamination above EU standards exceeded 90 percent in both study regions, whereas one year later, on the same farms, they found contamination rates of 6 percent in one region and 10 in the other. Given the significant spatial variation in aflatoxin levels among control farmers in our sample, documented above in Section 4, the small average effect may disguise heterogeneous effects.

We use spatial data on agro-climatic conditions during the growing season to estimate each village's average risk of aflatoxin contamination in the absence of the treatment. In line with previous work by Bowen and Hagan (2015), we identify two key predictors of aflatoxin contamination: dry spell incidence and average max temperature during the growing season. Using these two predictors, we proceed to estimate climate-induced aflatoxin risk in three steps. First, we estimate the relationship between the natural log of observed aflatoxin levels in control villages and our agro-climatic variables using LASSO and a simpler quadratic specification.<sup>41</sup> Second, using this estimated relationship between temperature, precipitation, and contamination in control villages, we predict village-level aflatoxin contamination risk for all villages in our sample. Third, we generate a dummy variable for predicted "high-risk" villages where predicted average contamination exceeds EU phytosanitary standards. Columns (2) and (3) of Table 3 show that we find consistently that the treatment effect is statistically significant and large in magnitude in villages predicted to be at high risk of aflatoxin contamination, and indistinguishable from zero in low-risk villages.<sup>42</sup>

 $<sup>^{41}</sup>$ Results are robust to a variety of alternative specifications, such as including temperature or dry season bins. Results are additionally robust to including or excluding baseline farmer characteristics, including lead farmer status and baseline aflatoxin awareness. We observe virtually no adoption of complementary practices for aflatoxin prevention at baseline, such as drying crops on a tarp (0%) or storing crops in hermetic bags (2%), so we do not include these variables in this step.

 $<sup>^{42}</sup>$ We observe similar patterns when we simply interact treatment with spatial fixed effects at the cooperative or commune level. Endline survey data on farmer practices suggests that differences in this treatment effect are not driven by differences in farmer practices. Table C.2 shows that there are no detectable differences in endline adoption of improved drying practices or improved storage technologies. There is also no detectable heterogeneity in Aflasafe adoption by predicted risk, suggesting farmers did not adjust their adoption behavior

	(1)	(2)	(3)
Panel A: ITT estimates			
Treated	0.08	-0.03	-0.08
	(0.050)	(0.060)	(0.070)
Treated $\times$		0.33***	
Pred. high risk (LASSO)		(0.110)	
Treated $\times$			0.25**
Pred. high risk (quadratic)			(0.130)
Pred. high risk (LASSO)		-0.34***	
		(0.090)	
Pred. high risk (quadratic)			-0.39***
			(0.080)
Panel B: TOT estimates (add	nin data ad	loption)	
Adopted	0.10	-0.05	-0.11
	(0.070)	(0.080)	(1.260)
Adopted $\times$		0.38***	
Pred. high risk (LASSO)		(0.130)	
Adopted $\times$			0.33
Pred. high risk (quadratic)			(1.280)
Pred. high risk (LASSO)		-0.39***	
		(0.100)	
Pred. high risk (quadratic)			-0.39***
_ 、_ ,			(0.080)
N	328	328	328
Control mean	0.659	0.659	0.659

Table 3: Phytosanitary standard compliance

This table shows results of regressions where the outcome variable is a dummy equal to one if the groundnut sample complied with EU phytosanitary standards. Panel A presents Intention-To-Treat results where the outcome is regressed on treatment status. Panel B presents 2SLS results where adoption (as measured in cooperative administrative data) is instrumented by treatment status. Columns (2) and (3) include an interaction with a dummy equal to one if the village was predicted to be at high risk given agro-climatic conditions experienced during the growing season. See Appendix C for details. In panel B, the 2SLS regression additionally includes treatment interacted with the predicted high risk dummy as a second instrument. Standard errors are clustered at the treatment assignment (village) level and bootstrapped in columns (2) and (3). See Table B.3 for a longer table including an additional measure of adoption and estimates including additional baseline controls.

in ex-ante anticipation of this risk. We also do not detect any significant relationship between climate-induced contamination risk and baseline aflatoxin awareness. When we consider sample attrition due to missing groundnut samples for aflatoxin testing, there is no differential attrition by predicted risk.

As discussed above in Section 4, statistical analysis of the continuous measure of quality is complicated by the fact that 49 percent of test results returned an aflatoxin level below the minimum level of detection of the testing equipment (2 parts per billion). We show results in Table B.4, but caution the reader that these results may not be robust to alternative methods of addressing the censored nature of the outcome. Nevertheless, the story is quite similar to the binary measure which suffers no such problem. We are again unable to detect a significant average treatment effect, although the point estimate is negative (suggesting lower contamination levels among farmers in the treatment group). When we allow the effect to differ by predicted contamination risk, we again find that the contract had a large and significant impact on quality in high-risk villages.

How does this result compare to other studies of quality upgrading? Magnan et al. (2021) is the closest comparable study, in which they offer farmers a low-cost technology (a tarp for drying groundnuts) and a price premium for improved quality. They find treatment effects of technology provision on EU standard compliance of 40-46%, with similar but less precise effects of a price premium.<sup>43</sup> If we define an analogous outcome, we find treatment effects on EU standard compliance of 22-57%.<sup>44</sup>

More broadly, this finding demonstrates an avenue for smallholder farmers to comply with international standards and access lucrative export markets. Compliance with phytosanitary standards is often quite costly for farmers (Asfaw et al., 2009), and changing standards have played a significant role in stymieing past attempts to link smallholders to export markets (Ashraf et al., 2009). We find that using a relatively cheap new technology—Aflasafe—farmers can significantly increase their likelihood of standards compliance when they otherwise would be at high risk of non-compliance. However, while necessary for export market access, standards compliance is not sufficient. Exporters are typically unwilling to work directly with small farmers. This underscores the importance of aggregating high-quality output, which we discuss in Section 6.4 below.

#### 6.2.1 Robustness to missing aflatoxin data

As discussed above in Section 4, we face non-trivial attrition in our quality measure, with only 83% of farmers providing a sample for testing. In follow-ups, we collected data which explains some of the attrition, namely that about 15% of our sample harvested either zero groundnuts

<sup>&</sup>lt;sup>43</sup>Other work has focused on maize, another important crop in sub-Saharan Africa commonly affected by aflatoxin contamination. EU standards for aflatoxin in maize are slightly higher at 10ppb. Contamination risks may also differ from groundnuts. Providing farmers with training, hermetic storage bags, and drying tarps for maize has been found to increase EU standard compliance by 33-71% (Pretari et al., 2019; Bauchet et al., 2020).

<sup>&</sup>lt;sup>44</sup>Macchiavello and Miquel-Florensa (2019) consider a very different setting—coffee cultivation—and find that a contract farming program induced upgrading along a variety of dimensions, with treatment effects of 2-25% depending on the outcome. Similarly, Park et al. (2021) find that information provision to farmers and intermediaries increased compliance with international pesticide standards for dragonfruit production by 20-68%. In a non-agricultural context, Atkin et al. (2017) find linking rug-producing firms to export markets increases quality by 26%.

(i.e., total crop failure) or harvested a quantity that falls below the median baseline quantity kept for seeds and home consumption. However, we also observe a differentially greater probability of attrition in the treatment group. As such, we implement two strategies to test the robustness of our quality findings to this non-random attrition.

	Main Spec	Horowitz Upper	& Manski Lower	Multiple Imputation
Treated	-0.03 (0.060)	-0.22 (0.060)	$0.08 \\ (0.050)$	-0.07 (0.060)
Treated $\times$ Pred. high risk (LASSO)	0.33 (0.110)	0.24 (0.110)	$0.35 \\ (0.080)$	$0.29 \\ (0.090)$
Pred. high risk (LASSO)	-0.34 (0.090)	-0.26 (0.080)	-0.34 (0.080)	-0.17 (0.100)
N Control mean	$\begin{array}{c} 328 \\ 0.659 \end{array}$	$396 \\ 0.701$	$396 \\ 0.579$	$\frac{396}{0.619}$

Table 4: Phytosanitary standard compliance and robustness to sample attrition

This table shows results of regressions where the outcome variable is a dummy equal to one if the groundnut sample complied with EU phytosanitary standards. Column 1 replicates the results shown in column 2 of Table 3. Columns 2 and 3 show the upper and lower Horowitz and Manski (2000) bounds which imposed no distributional assumptions on missing data. Column 4 shows multiply-imputed parameter estimates with 50 imputations, where missing test results are imputed using a linear regression with baseline controls. Standard errors are clustered at the treatment assignment (village) level and bootstrapped in columns 1-3.

Table 4 presents the results of these strategies. In columns 2 and 3, we show the upper and lower Horowitz and Manski (2000) bounds. More specifically, we estimate the upper bound by replacing missing values for treated farmers with the upper bound of the unconditional distribution and replacing missing values for control farmers with the lower bound of the unconditional distribution. The lower bound is analogous but uses the unconditional minimum for missing treated farmers and the unconditional maximum for control farmers. In column 4, we conduct a multiple imputation exercise where we first predict missing test result values using baseline farmer characteristics. We bootstrap the imputation exercise 50 times and report the average estimated parameters. These exercises should provide some reassurance that our findings on quality improvements in high-risk areas are not driven by missing data.

#### 6.3 Technology adoption two years later

Two years after the conclusion of the original study, our two partner cooperatives decided to make Aflasafe widely available to their members.<sup>45</sup> The cooperatives first held information sessions for village section presidents, in which representatives from the manufacturer and

<sup>&</sup>lt;sup>45</sup>The COVID-19 crisis significantly constrained cooperatives, and the groundnut sector in general, during the year following the original study.

IITA described the problem of affatoxin contamination and how Affasafe could improve crop quality outcomes. Village section presidents were asked in turn to share information with their members.

Following these information sessions, farmers were invited to submit expressions of interest to the cooperative. This process functioned as it normally does for requesting seeds and fertilizer, but with the inclusion of Aflasafe in input packages. After receiving these expressions, the cooperatives decided how to allocate credit for Aflasafe in the same way they allocate credit for other inputs. Cooperative representatives state this is a function of perceived reliability and total credit available. Aflasafe was sold exclusively on credit. Cooperative representatives promised to send agents to train farmers when Aflasafe was distributed. Farmers were not promised a specific price premium for aflatoxin standard compliance, but may have harbored beliefs that a premium would be available. Thus, the conditions during the follow-up season partially matched the conditions of our bundled contracting treatment, but without a price premium guarantee.

In this section, we consider two outcomes from cooperative administrative data: did a farmer express interest in purchasing Aflasafe, and did they purchase it? The first outcome is purely dependent on farmers, and is not contingent on credit availability from the cooperative. The latter outcome is determined jointly by farmers' interest and cooperative staff. It is instructive to consider both outcomes, since past treatment status may have influenced cooperatives' perceptions of a given farmer's reliability or capability using Aflasafe. In practice, the two outcomes largely overlap.

For our outcomes of interest in this section, we modify equation 1 above to estimate the following equation via OLS:

$$Y_{ijk} = \alpha + \beta_1 T_{jk} + \beta_2 P_{ijk} + \beta_3 (T_{jk} \times P_{ijk}) + \beta_4 S_{jk} + \gamma_k + \epsilon_{ijk}$$
(3)

where  $T_{jk}$  is the treatment assignment of village j in commune k. In some columns we add  $P_{ijk}$ , a dummy equal to one if farmer i was sampled for inclusion in the previous study, and interact it with  $T_{jk}$ . We also sometimes include  $S_{jk}$ , a dummy equal to one if a village was sampled for inclusion in the original study.  $\gamma_k$  is a commune fixed effect, and standard errors  $\epsilon_{ijk}$  are clustered at the village level.

#### 6.3.1 Direct and Indirect Effects

First, we turn to the direct effects of past assignment to the contract treatment. Table 5 shows the effects of treatment assignment on interest and purchase decisions. In the first two columns, we restrict the sample to past study participants. It is worth noting first that average adoption among farmers in previously-treated villages is significantly lower than the adoption we observed in the original experiment. Patterns of technology adoption and disadoption are common in the literature (Suri, 2011; Nourani, 2019), and perhaps

are particularly unsurprising in a context like ours where farmers must learn about the effectiveness of a technology in reducing a contamination which is costly to observe and for which contamination risk is stochastic. The lingering effects of the COVID-19 crisis may have also reduced farmer willingness to experiment. Despite this substantially lower adoption, we nevertheless find that among past study participants, treated farmers are significantly more likely to be interested in purchasing Aflasafe, and more likely to actually purchase it in 2021.

	Parti	cipants	Study	Villages	All Villages	
	(1)	(2)	(3)	(4)	(5)	(6)
	Int.	Bought	Int.	Bought	Int.	Bought
Treated Village in	0.08**	0.09**	0.01	0.02	-0.01	0.00
2019	(0.03)	(0.04)	(0.02)	(0.02)	(0.03)	(0.02)
Participant in 2019			-0.03	-0.01	-0.03	-0.01
			(0.02)	(0.02)	(0.02)	(0.02)
Treated Village in			$0.07^{*}$	$0.07^{*}$	0.08**	0.08**
2019 $\times$ Participated in 2019			(0.04)	(0.04)	(0.04)	(0.04)
Study Village in					-0.02	-0.02
2019					(0.02)	(0.02)
Observations	392	392	1335	1335	4466	4466
$R^2$	0.098	0.095	0.062	0.040	0.083	0.086
Excluded Group Mean Dep. Var	0.05	0.05	0.09	0.07	0.12	0.11

Table 5: Aflasafe adoption in 2021

Results in this table are from linear regressions of dummies equal to one if a farmer expressed interest in, or actually purchased, Aflasafe in 2021. Odd columns report stated interest in purchasing Aflasafe, as reported to cooperatives, and even columns report actual purchases. These outcomes are measured using cooperative administrative data. The first two columns restrict only to past study participants. The third and fourth columns expand the sample to include non-participants in study villages. The fifth and sixth columns include all cooperative members. Standard errors (in parentheses) are clustered at the treatment assignment (village) level. All regressions include commune fixed effects.

We can additionally test for the presence of within-village spillovers in 2021 adoption decisions. In addition to the randomly-assigned 2019 treatment status of villages, we additionally exploit the random selection of study participants from the larger pool of cooperative members. As we expand the sample to include non-participants from study villages in the third and fourth columns, we see that the increase in purchases is driven almost entirely by past participants themselves, with little evidence of within-village spillovers to non-participants.

Finally, we can also exploit the random selection of villages for inclusion in the original study. In the fifth and sixth columns of Table 5, we find little evidence of information effects distinct from past treatment. Farmers in non-treated study villages received information about Aflasafe, but with few actual adopters in 2019. Somewhat in contrast to Treurniet (2021), we find no evidence that participation status in the 2019 study, separately from treatment status, has any impact on adoption two years later. Thus, it appears that information alone in 2019 was insufficient to increase adoption in 2021. This is despite evidence that farmers in this group were more aware of aflatoxins in general.<sup>46</sup>

#### 6.3.2 Experiment Participants

If we restrict the sample to past participants, for whom we collected baseline characteristics in 2019, we can also learn more who benefits the most from past experience. Mirroring analyses in previous sections, we consider several dimensions of heterogeneity according to behavioral characteristics like risk and reciprocity.

In Table 6, we observe significant heterogeneity in follow-up adoption along two behavioral dimensions: reciprocity and risk aversion. Consistent with our results on adoption in 2019 (see Table B.1), we find that the treatment effect is larger among highly-reciprocal farmers, but the effect is not statistically significant. It may be that reciprocity continues to play a role, or simply that past experience is higher among these farmers leading to higher adoption in the current season. Similarly, we see in 2019 that highly risk-loving farmers are less affected by the treatment, and more likely to adopt in absence of the treatment. This pattern carries through to 2021, albeit relatively more strongly. It may be that past experience via the treatment enables more risk-averse farmers to invest in quality upgrading.

#### 6.4 Farmer commercialization decisions

Finally, we consider the impact of the intervention on farmer commercialization decisions. Namely, we consider whether the intervention is sufficient to allow intermediaries (in our case, cooperatives) to aggregate farmers' higher-quality production. This is an important outcome for policy (Barrett, 2008; Fischer and Qaim, 2012), especially given the well-documented challenges cooperatives face aggregating farmer output (Bernard et al., 2008, 2015; Aflagah et al., 2022). If the intervention succeeds at this, cooperatives could plausibly offer our intervention at scale and induce widespread improvements in production quality and food safety. This outcome is also of academic interest. Given that the treatment represents a modification of the relational contract between farmers and cooperatives, it is useful to measure to what extent this change improves the relationship by increasing output sales to the cooperative. In particular, given the high spot market prices in the season we study, the outcomes we observe here demonstrate the role of our intervention in improving informally-contracted relationships when the temptation to cheat is high.

Column 1 of Table 7 shows the effect of the contract offer on output sales to the cooperative at the extensive margin. The outcome is a dummy equal to one if the farmer reported any output sales to the cooperative at endline. On average, treatment increased

<sup>&</sup>lt;sup>46</sup>Results from a phone survey conducted in May and June 2020 with study participants and a random sample of non-participants, available upon request.

the probability of selling any output to the cooperative by 12 percentage points. Given the low rate at which farmers decide to sell output to cooperatives, this represents a sizeable change in behavior at the extensive margin. Results analyzing the intensive margin of output allocation, shown in Table B.6, tell a similar story.<sup>47</sup>

How do these findings compare to the literature? Aflagah et al. (2022) find that in larger village cooperative groups, a cheap talk intervention of sharing ex-ante collective commercialization intentions increased the probability of sales to the cooperative by 1 percentage point (13%) per group member. At the median group size (24 members), their intervention increases collective commercialization by 24 percentage points. By comparison, our point estimate suggests a 12 percentage point (52%) increase in the probability of sales to the cooperative.

<sup>&</sup>lt;sup>47</sup>One interpretation of this intensive-margin result is that treated farmers simply sold enough extra output to the cooperative to repay their credit for purchasing Aflasafe. However, the magnitude of the average treatment effect is roughly double that of the cost of Aflasafe. Indeed this increase in output aggregation is sufficient to cover all of a cooperative's marginal costs if they were additionally paying for aflatoxin testing. Combined with the premium buyers are willing to pay cooperatives for certified high-quality production, we estimate our partner cooperatives could at least break even if they scaled the intervention to all of their members and those members increased output sales to the cooperative by a similar margin.

	(1)	(2)	(3)	(4)
Panel A: Interested in Purcha	sing Aflase	afe		
Treated	0.10***	0.08**	0.03	0.14***
	(0.030)	(0.040)	(0.060)	(0.040)
Treated $\times$		0.06		
Reciprocal		(0.070)		
Treated $\times$			0.10	
Patient			(0.070)	
Treated $\times$				-0.17***
Risk-loving				(0.050)
Reciprocal	-0.03	-0.06*	-0.03	-0.03
	(0.040)	(0.030)	(0.040)	(0.040)
Patient	-0.03	-0.03	-0.08**	-0.02
	(0.040)	(0.040)	(0.030)	(0.040)
Risk-loving	-0.01	-0.01	-0.01	$0.08^{*}$
5	(0.030)	(0.030)	(0.030)	(0.040)
N	392	392	392	392
Control mean	0.046	0.046	0.046	0.046
Panel B: Purchased Aflasafe				
Treated	0.11***	0.08**	0.03	$0.15^{***}$
	(0.030)	(0.040)	(0.060)	(0.040)
Treated $\times$		0.08		
Reciprocal		(0.070)		
Treated $\times$			0.10	
Patient			(0.070)	
Treated $\times$				-0.17***
Risk-loving				(0.050)
Reciprocal	-0.04	-0.08***	-0.04	-0.04
	(0.040)	(0.020)	(0.040)	(0.040)
Patient	-0.03	-0.03	-0 09**	-0.03
	(0.040)	(0.040)	(0.030)	(0.040)
Risk-loving	-0.01	-0.01	-0.01	0.09*
	(0.030)	(0.030)	(0.030)	(0.040)
N	392	392	392	392
Control mean	0.041	0.041	0.041	0.041

Table 6: Adoption of Aflasafe two years after original study, with behavioral heterogeneity

This table shows results of regressions where the outcome variables are shown at the top of each panel. Panel A presents Intention-To-Treat results where the outcome is a dummy equal to one if a farmer indicated binding interest in purchasing Aflasafe from the cooperative. Panel B presents Intention-To-Treat results where the outcome is a dummy equal to one if a farmer purchased Aflasafe from the cooperative. Column (2) additionally includes an interaction with a dummy equal to one if self-assessed intrinsic reciprocity is greater or equal to 6 on a 10 point scale, column (3) includes an interaction with a dummy equal to one for self-assessed patient farmers, and column (4) includes an interaction with a dummy equal to one if the farmer identified as extremely willing to take risks. All regressions include commune fixed effects to account for stratiged randomization of treatment assignment. Standard errors are clustered at the treatment assignment (village) level.

	(1)	(2)	(3)	(4)
Panel A: ITT estimates				
Treated	$0.12^{**}$	0.04	0.00	$0.15^{**}$
	(0.050)	(0.050)	(0.080)	(0.060)
Treated $\times$		$0.27^{***}$		
Reciprocal		(0.090)		
Treated $\times$			$0.16^{*}$	
Patient			(0.090)	
Treated $\times$				-0.10
Risk loving				(0.090)
Reciprocal	-0.02	-0.17***	-0.03	-0.02
	(0.050)	(0.050)	(0.050)	(0.050)
Patient	0.00	-0.01	-0.08	0.00
	(0.050)	(0.050)	(0.070)	(0.050)
Risk loving	0.11**	0.11**	0.11*	0.17**
-	(0.050)	(0.050)	(0.050)	(0.070)
Panel B: TOT estimates (admi	n data adop	tion)		
Adopted	0.15***	0.05	-0.01	$0.18^{***}$
	(0.060)	(0.070)	(0.110)	(0.060)
Adopted $\times$		0.31***		
Reciprocal		(0.110)		
Adopted $\times$			$0.21^{*}$	
Patient			(0.120)	
Adopted $\times$				-0.12
Risk loving				(0.130)
Reciprocal	-0.03	-0.21***	-0.04	-0.03
	(0.050)	(0.070)	(0.050)	(0.050)
Patient	0.00	-0.01	-0.11	0.01
	(0.050)	(0.050)	(0.080)	(0.050)
Risk loving	$0.10^{*}$	0.11**	0.10*	0.17**
-	(0.050)	(0.050)	(0.050)	(0.090)
N	370	370	370	370
Control mean	0.230	0.230	0.230	0.230

Table 7: Sold any output to coop, with behavioral heterogeneity

This table shows results of regressions where the outcome variable is a dummy equal to one if the farmer reported selling any output to the cooperative. Panel A presents Intention-To-Treat results where the outcome is regressed on treatment status. Panel B presents 2SLS results where adoption (as measured in cooperative administrative data) is instrumented by treatment status. Column (2) additionally includes an interaction with a dummy equal to one if self-assessed intrinsic reciprocity is greater or equal to 0.6 on a standardized [0,1] scale, column (3) includes an interaction with a dummy equal to one for self-assessed patient farmers, and column (4) includes an interaction with a dummy equal to one if the farmer identified as extremely willing to take risks. In panel B, the 2SLS regressions additionally include treatment interacted with the cooperative dummy as a second instrument. All regressions include commune fixed effects to account for stratified randomization of treatment assignment. Standard errors are clustered at the treatment assignment (village) level. See Table B.5 for corresponding results using an additional measure of adoption and with additional baseline controls included.

#### 6.4.1 Behavioral Mechanisms

Behavioral characteristics like reciprocity and patience can play important roles in the success of informal relationships (Leider et al., 2009; Finan and Schechter, 2012; Ligon and Schechter, 2012). Relationships between farmers and cooperatives are not purely commercial, as they also invoke a sense of collective action and solidarity among members. Understanding the behavioral mechanisms that drive increased collective commercialization could improve contract design or targeting. Table 7 shows that there is a significant and positive interaction between the treatment and a baseline measure of intrinsic reciprocity. The marginal effect of treatment is small or statistically indistinguishable from zero for low-reciprocity farmers. By comparison, for the most reciprocal farmers, the treatment increases output allocation to the cooperative by about 27 percentage points.

Patience is also a key component of improving collective commercialization. Farmers who sell output to traders typically get paid immediately, in cash. By comparison, sales via the cooperative often involve a delay receiving some or all of the proceeds from the sale. The new contract may require additional patience, as quality-contingent premium payments require farmers to wait for results from an aflatoxin test. Column (3) of Table 7 demonstrates that the treatment effect was indeed larger for more patient farmers, although the effect is imprecise. These farmers may have been more willing to wait for the potential rewards associated with allocating output to the cooperative. The wording of the patience elicitation question are also suggestive: we asked farmers how they assessed their willingness to give up something today in order to benefit in the future. Given that cooperatives may allocate access to new technologies and trials based on perceived reliability, it could be that the farmers we flag as more patient recognize this and respond accordingly.

By contrast, we fail to detect any evidence of heterogeneity by risk aversion. We observe a highly bimodal distribution of self-assessed risk aversion on an 11-point scale. This matches closely with earlier work by Charness and Viceisza (2016) in a similar context. We flag self-identified "risk-loving" farmers as those who responded that their willingness to take risks was 10 out of 10. Failure to detect an outcome could indicate a need to better adapt risk measurement methods to our precise context. Alternatively, it could indicate that risk aversion does not impact the decision to sell output to the cooperative. This seems less likely, given that farmers who choose to sell output elsewhere risk reduced access to future benefits from the cooperative.

## 7 Concluding Remarks

In this paper, we present new evidence that a simple, flexible contracting arrangement can induce improvements in crop quality via the adoption of a new technology. Farmers who adopt this technology in areas where quality would otherwise be low produce significantly higher-quality groundnuts. The contracting arrangement improves commercial outcomes at cooperatives, with treatment effects concentrated among farmers who are more reciprocal and patient. The contract increases technology adoption two years after the study, but there is significant disadoption by previously treated farmers. These findings have important implications for contract design, particularly in environments with low enforcement capabilities and where contracts involve using previously-unknown technologies or techniques.

Our work suggests several avenues for further research. In particular, although the bundled contract was highly effective at increasing Aflasafe adoption and did improve quality in high-risk areas, implementation of a comprehensive contract is costly. Many buyers are unwilling to commit to a fixed premium up front. Could a simpler resource-providing contract without price guarantees provide similar benefits? This could be particularly relevant given the repeated game nature of the interaction between farmers and cooperatves. Producers may not need guarantees of training or price rewards as their experience with the technology increases and they learn about potential market rewards to quality upgrading. Our follow-up results suggest the price premium guarantee may play an important role in determining farmers' decisions to invest in quality upgrading, in line with Hoffmann et al. (2022).

Additionally, our finding that the contract induced changes in commercialization decisions by producers merits further study. Given the well-documented difficulties cooperatives face in effectively aggregating output (Aflagah et al., 2022), our results suggest an avenue for improving cooperative performance by focusing on supporting higher-quality production. Future work could delve further into the behavioral heterogeneity we observe, to better understand how factors like reciprocity and reputation impact producer decisions.

## References

- Abate, G. T., Bernard, T., de Brauw, A., and Minot, N. (2018). The impact of the use of new technologies on farmers' wheat yield in Ethiopia: Evidence from a randomized control trial. Agricultural Economics, 49(4):409–421.
- Aflagah, K., Bernard, T., and Viceisza, A. (2022). Cheap talk and coordination in the lab and in the field: Collective commercialization in Senegal. *Journal of Development Economics*, 154:102751.
- Agyekum, M. and Jolly, C. M. (2017). Peanut trade and aflatoxin standards in Europe: Economic effects on trading countries. *Journal of Policy Modeling*, 39(1):114–128.
- Anderson, W. F., Holbrook, C. C., Wilson, D. M., and Matheron, M. E. (1995). Evaluation of preharvest aflatoxin contamination in several potentially resistant peanut genotypes. *Peanut Science*, 22(1):29–32.
- Angelucci, M., Karlan, D., and Zinman, J. (2015). Microcredit impacts: Evidence from a randomized microcredit program placement experiment by Compartamos Banco. American Economic Journal: Applied Economics, 7(1):151–182.
- Arouna, A., Michler, J. D., and Lokossou, J. C. (2021). Contract farming and rural transformation: Evidence from a field experiment in Benin. *Journal of Development Economics*, 151:102626.
- Asfaw, S., Mithöfer, D., and Waibel, H. (2009). Investment in compliance with GlobalGAP standards: Does it pay off for small-scale producers in Kenya? Quarterly Journal of International Agriculture, 48(4):337–362.
- Ashraf, N., Giné, X., and Karlan, D. (2009). Finding missing markets (and a disturbing epilogue): Evidence from an export crop adoption and marketing intervention in Kenya. *American Journal of Agricultural Economics*, 91(4):973–990.
- Atkin, D., Khandelwal, A. K., and Osman, A. (2017). Exporting and firm performance: Evidence from a randomized experiment. *The Quarterly Journal of Economics*, 132(2):551– 615.
- Bai, J. (2018). Melons as lemons: Asymmetric information, consumer learning and quality provision. Working Paper.
- Bandyopadhyay, R., Atehnkeng, J., Ortega-Beltran, A., Akande, A., Falade, T. D. O., and Cotty, P. J. (2019). "Ground-truthing" efficacy of biological control for aflatoxin mitigation in farmers' fields in Nigeria: From field trials to commercial usage, a 10-year study. *Frontiers in Microbiology*, 10.
- Barrett, C. B. (2008). Smallholder market participation: Concepts and evidence from eastern and southern Africa. *Food Policy*, 33(4):299–317.
- Barrett, C. B., Reardon, T., Swinnen, J., and Zilberman, D. (2020). Agri-food value chain revolutions in low-and middle-income countries. *Journal of Economic Literature*, Forthcoming:68.

- Bauchet, J., Prieto, S., and Ricker-Gilbert, J. (2020). Improved drying and storage practices that reduce aflatoxins in stored maize: Experimental evidence from smallholders in Senegal. *American Journal of Agricultural Economics*, Forthcoming:ajae.12106.
- Beaman, L., Karlan, D., Thuysbaert, B., and Udry, C. (2020). Selection into credit markets: Evidence from agriculture in Mali. Working Paper.
- Bellemare, M. F., Lee, Y. N., and Novak, L. (2021). Contract farming as partial insurance. World Development, 140:84.
- Bellemare, M. F. and Novak, L. (2017). Contract farming and food security. American Journal of Agricultural Economics, 99(2):357–378.
- Bernard, T., Collion, M.-H., de Janvry, A., Rondot, P., and Sadoulet, E. (2008). Do village organizations make a difference in African rural development? A study for Senegal and Burkina Faso. World Development, 36(11):2188–2204.
- Bernard, T., de Janvry, A., Mbaye, S., and Sadoulet, E. (2017). Expected product market reforms and technology adoption by Senegalese onion producers. *American Journal of Agricultural Economics*, 99(4):1096–1115.
- Bernard, T., Sene, L., Wouterse, F., and Viceisza, A. (2015). Leaders needed: Experimental evidence from rural producer organizations in Senegal. Working Paper.
- Blankson, G. K., Mills-Robertson, F. C., and Ofosu, I. W. (2019). Survey of occurrence levels of aflatoxins in selected locally processed cereal-based foods for human consumption from Ghana. *Food Control*, 95:170–175.
- Bold, T., Ghisolfi, S., Nsonzi, F., and Svensson, J. (2022). Market Access and Quality Upgrading: Evidence from Three Field Experiments. *American Economic Review*, page 72.
- Bowen, K. and Hagan, A. (2015). Temperature and moisture conditions that affect aflatoxin contamination of peanuts. *Peanut Science*, 42(2):121–127.
- Brooks, G. E. (1975). Peanuts and colonialism: Consequences of the commercialization of peanuts in West Africa, 1830-70. The Journal of African History, 16(1):29–54.
- Brown, M., Falk, A., and Fehr, E. (2004). Relational contracts and the nature of market interactions. *Econometrica*, 72(3):747–780.
- Bruhn, M. and McKenzie, D. (2009). In pursuit of balance: Randomization in practice in development field experiments. *American Economic Journal: Applied Economics*, 1(4):200–232.
- Bryden, W. L. (2012). Mycotoxin contamination of the feed supply chain: Implications for animal productivity and feed security. Animal Feed Science and Technology, 173(1):134– 158.
- Bulte, E., Cecchi, F., Lensink, R., Marr, A., and van Asseldonk, M. (2019). Does bundling crop insurance with certified seeds crowd-in investments? experimental evidence from Kenya. Journal of Economic Behavior & Organization, page S0167268119302264.
- Cabral, L., Ozbay, E. Y., and Schotter, A. (2014). Intrinsic and instrumental reciprocity: An experimental study. *Games and Economic Behavior*, 87:100–121.

- Casaburi, L. and Willis, J. (2018). Time versus state in insurance: Experimental evidence from contract farming in Kenya. *American Economic Review*, 108(12):3778–3813.
- Casswell, N. (1984). Autopsie de l'ONCAD: La politique arachidiere au Senegal, 1966-1980. *Politique Africaine*, 14.
- CBI (2022). The European market potential for groundnuts | CBI. https://www.cbi.eu/market-information/processed-fruit-vegetables-ediblenuts/groundnuts/market-potential.
- Charness, G. and Haruvy, E. (2002). Altruism, equity, and reciprocity in a gift-exchange experiment: An encompassing approach. *Games and Economic Behavior*, page 29.
- Charness, G. and Viceisza, A. (2016). Three risk-elicitation methods in the field: Evidence from rural Senegal. *Review of Behavioral Economics*, 3(2):145–171.
- Chowdhury, S., Smits, J., and Sun, Q. (2020). Contract structure, time preference, and technology adoption. Discussion Paper 13590, IZA Institute of Labor Economics.
- Clavel, D., da Sylva, A., Ndoye, O., and Mayeux, A. (2013). Amélioration de la qualité sanitaire de l'arachide au Sénégal : un challenge pour une opération de recherche-développement participative. *Cahiers Agricultures*, 22(3):174–181.
- Clavel, D. and Gaye, M. (2018). L'émergence de nouvelles coopératives semencières au Sénégal Analyse de l'impact de la recherche-développement sur l'arachide de 1999 à 2016. *Cahiers Agricultures*, 27(1):15008.
- Cotty, P. J. and Jaime-Garcia, R. (2007). Influences of climate on aflatoxin producing fungi and aflatoxin contamination. *International Journal of Food Microbiology*, 119(1):109–115.
- Coursaget, P., Depril, N., Chabaud, M., Nandi, R., Mayelo, V., LeCann, P., and Yvonnet, B. (1993). High prevalence of mutations at codon 249 of the p53 gene in hepatocellular carcinomas from Senegal. *British Journal of Cancer*, 67(6):1395–1397.
- Crépon, B., Devoto, F., Duflo, E., and Parienté, W. (2015). Estimating the impact of microcredit on those who take it up: Evidence from a randomized experiment in Morocco. *American Economic Journal: Applied Economics*, 7(1):123–150.
- de Janvry, A. and Sadoulet, E. (2020). Using agriculture for development: Supply- and demand-side approaches. *World Development*, 133:105003.
- Deutschmann, J. W., Duru, M., Siegal, K., and Tjernström, E. (2021). Can smallholder extension transform African agriculture? Working Paper 26054, National Bureau of Economic Research.
- Devoto, F., Duflo, E., Dupas, P., Parienté, W., and Pons, V. (2012). Happiness on tap: Piped water adoption in urban Morocco. *American Economic Journal: Economic Policy*, 4(4):68–99.
- Diagana, B. (2008). L'expérience du Comité national interprofessionnel de l'arachide. In Les Organisations Interprofessionelles Agricoles Au Sénégal : De Nouveaux Outils de Régulation Des Marchés. Bureau d'analyses macro-économiques de l'Institut sénégalais de recherches agricoles, Dakar.

- Dorner, J. W. (2009). Development of biocontrol technology to manage aflatoxin contamination in peanuts. *Peanut Science*, 36(1):60–67.
- Dorner, J. W. and Lamb, M. C. (2006). Development and commercial use of afla-Guard®, an aflatoxin biocontrol agent. *Mycotoxin Research*, 22(1):33–38.
- Doster, M. A., Cotty, P. J., and Michailides, T. J. (2014). Evaluation of the atoxigenic aspergillus flavus strain AF36 in pistachio orchards. *Plant Disease*, 98(7):948–956.
- Duflo, E., Kremer, M., and Robinson, J. (2011). Nudging farmers to use fertilizer: Theory and experimental evidence from Kenya. *American Economic Review*, 101(6):2350–2390.
- Eclosio (2018). Rapport d'activites 2018. Annual Report, Eclosio.
- EFSA (2007). Opinion of the scientific panel on contaminants in the food chain [CONTAM] related to the potential increase of consumer health risk by a possible increase of the existing maximum levels for aflatoxins in almonds, hazelnuts and pistachios and derived products. *EFSA Journal*, 5(3):446.
- Emerick, K. and Dar, M. H. (2020). Farmer field days and demonstrator selection for increasing technology adoption. *The Review of Economics and Statistics*, pages 1–41.
- Fahn, M. (2020). Reciprocity in dynamic employment relationships. Working Paper 8414, CESifo.
- Falk, A., Becker, A., Dohmen, T., Enke, B., Huffman, D., and Sunde, U. (2018). Global evidence on economic preferences. *The Quarterly Journal of Economics*, 133(4):1645–1692.
- Falk, A., Becker, A., Dohmen, T. J., Huffman, D., and Sunde, U. (2016). The preference survey module: A validated instrument for measuring risk, time, and social preferences. SSRN Electronic Journal.
- Feder, G., Just, R. E., and Zilberman, D. (1985). Adoption of agricultural innovations in developing countries: A survey. *Economic Development and Cultural Change*, 33(2):255– 298.
- Fehr, E., Brown, M., and Zehnder, C. (2009). On reputation: A microfoundation of contract enforcement and price rigidity. *The Economic Journal*, 119(536):333–353.
- Fernandes, A. M., Ferro, E., and Wilson, J. S. (2019). Product standards and firms' export decisions. The World Bank Economic Review, 33(2):353–374.
- Ferro, E., Otsuki, T., and Wilson, J. S. (2015). The effect of product standards on agricultural exports. *Food Policy*, 50:68–79.
- Finan, F. and Schechter, L. (2012). Vote-buying and reciprocity. *Econometrica*, 80(2):863– 881.
- Fischer, E. and Qaim, M. (2012). Linking smallholders to markets: Determinants and impacts of farmer collective action in Kenya. *World Development*, 40(6):1255–1268.
- Florkowski, W. J. (2014). Strategies to control aflatoxin in groundnut value chains. Discussion Paper 01369, IFPRI.

- Fofana, I., Badiane, O., Camara, A., and Goundan, A. (2018). Groundnuts export tax in Senegal: Winners and losers. Discussion Paper 01757, IFPRI.
- Fontagné, L., Orefice, G., Piermartini, R., and Rocha, N. (2015). Product standards and margins of trade: Firm-level evidence. *Journal of International Economics*, 97(1):29–44.
- Foster, A. D. and Rosenzweig, M. R. (2010). Microeconomics of technology adoption. Annual Review of Economics, 2(1):395–424.
- Frisvad, J. C., Hubka, V., Ezekiel, C. N., Hong, S. B., Nováková, A., Chen, A. J., Arzanlou, M., Larsen, T. O., Sklenář, F., Mahakarnchanakul, W., Samson, R. A., and Houbraken, J. (2019). Taxonomy of Aspergillus section Flavi and their production of aflatoxins, ochratoxins and other mycotoxins. *Studies in Mycology*, 93:1–63.
- Garcia-Alvarez-Coque, J.-M., Taghouti, I., and Martinez-Gomez, V. (2020). Changes in aflatoxin standards: Implications for EU border controls of nut imports. *Applied Economic Perspectives and Policy*, page aepp/ppy036.
- Gnonlonfin, G. J. B., Hell, K., Adjovi, Y., Fandohan, P., Koudande, D. O., Mensah, G. A., Sanni, A., and Brimer, L. (2013). A review on aflatoxin contamination and its implications in the developing world: A sub-Saharan African perspective. *Critical Reviews in Food Science and Nutrition*, 53(4):349–365.
- Hansen, B. B. and Bowers, J. (2008). Covariate balance in simple, stratified and clustered comparative studies. *Statistical Science*, 23(2):219–236.
- Hendrix, M. C., Obed, I. L., Alice, M. M., Elijah, P., Jones, Y., Samuel, M. C. N., Rick, L. B., and David, J. (2019). Predicting aflatoxin content in peanuts using ambient temperature, soil temperature and soil moisture content during pod development. *African Journal of Plant Science*, 13(3):59–69.
- Heß, S. (2017). Randomization inference with Stata: A guide and software. The Stata Journal: Promoting communications on statistics and Stata, 17(3):630–651.
- Hoffmann, V., Jones, K., and Leroy, J. L. (2018a). The impact of reducing dietary aflatoxin exposure on child linear growth: A cluster randomised controlled trial in Kenya. *BMJ Global Health*, 3(6):e000983.
- Hoffmann, V., Kariuki, S., Pieters, J., and Treurniet, M. (2018b). Can markets support smallholder adoption of a food safety technology? Aflasafe in Kenya. Project Note, IFPRI.
- Hoffmann, V., Kariuki, S., Pieters, J., and Treurniet, M. (2022). Upside Risk, Consumption Value, and Market Returns to Food Safety. *American Journal of Agricultural Economics*.
- Hoffmann, V., Moser, C. M., and Herrman, T. J. (2020a). Demand for Aflatoxin-Safe maize in Kenya: Dynamic response to price and advertising. *American Journal of Agricultural Economics*, Forthcoming:ajae.12093.
- Hoffmann, V., Mutiga, S. K., Harvey, J. W., Nelson, R. J., and Milgroom, M. G. (2020b). Observability of food safety losses in maize: Evidence from Kenya. *Food Policy*, Forthcoming:101895.

- Holbrook, C. C., Kvien, C. K., Rucker, K. S., Wilson, D. M., Hook, J. E., and Matheron, M. E. (2000). Preharvest aflatoxin contamination in drought-tolerant and drought-intolerant peanut genotypes. *Peanut Science*, 27(2):45–48.
- Horowitz, J. L. and Manski, C. F. (2000). Nonparametric Analysis of Randomized Experiments with Missing Covariate and Outcome Data. *Journal of the American Statistical* Association, 95(449):77–84.
- IARC (1993). Some Naturally Occurring Substances: Food Items and Constituents, Heterocyclic Aromatic Amines and Mycotoxins.
- IARC (2012). Chemical Agents and Related Occupations, volume 100 of IARC Monographs on the Evaluation of Carcinogenic Risks to Humans. World Health Organization.
- Imai, K., King, G., and Stuart, E. A. (2008). Misunderstandings between experimentalists and observationalists about causal inference. *Journal of the Royal Statistical Society: Series A (Statistics in Society)*, 171(2):481–502.
- J-PAL and CEGA (2022). Evidence for Digital and Bundled Services: Framing a Research Agenda for the Digital Agricultural Innovations and Services Initiative. Technical report.
- Jordan, D., Appaw, W., Ellis, W. O., Akromah, R., Mochiah, M. B., Abudulai, M., Brandenburg, R. L., Jelliffe, J., Bravo-Ureta, B., and Boote, K. (2020). Evaluating improved management practices to minimize aflatoxin contamination in the field, during drying, and in storage in Ghana. *Peanut Science*, pages PS20–3.1.
- Kamala, A., Shirima, C., Jani, B., Bakari, M., Sillo, H., Rusibamayila, N., De Saeger, S., Kimanya, M., Gong, Y., and Simba, A. (2018). Outbreak of an acute aflatoxicosis in Tanzania during 2016. World Mycotoxin Journal, 11(3):311–320.
- Karlan, D., Kutsoati, E., McMillan, M., and Udry, C. (2011). Crop price indemnified loans for farmers: A pilot experiment in rural Ghana. *Journal of Risk and Insurance*, 78(1):37–55.
- Karlan, D., Osei, R., Osei-Akoto, I., and Udry, C. (2014). Agricultural decisions after relaxing credit and risk constraints. *The Quarterly Journal of Economics*, 129(2):597–652.
- Kelly, V. A., Diagana, B. N., Reardon, T., Gaye, M., and Crawford, E. W. (1996). Cash crop and foodgrain productivity in Senegal: Historical view, new survey evidence, and policy implications. Technical Report 54051, Michigan State University, Department of Agricultural, Food, and Resource Economics.
- Kessler, J. B. and Leider, S. (2012). Norms and contracting. Management Science, 58(1):62– 77.
- Kinyungu, S., Isakeit, T., Ojiambo, P. S., and Woloshuk, C. P. (2019). Spread of Aspergillus flavus and aflatoxin accumulation in postharvested maize treated with biocontrol products. *Journal of Stored Products Research*, 84:101519.
- Leider, S., Möbius, M. M., Rosenblat, T., and Do, Q.-A. (2009). Directed altruism and enforced reciprocity in social networks. *Quarterly Journal of Economics*, 124(4):1815–1851.

- Lerva, B. (2020). Quantifying externalities in technology adoption: Experimental evidence from Ugandan farmers. Working Paper.
- Ligon, E. and Schechter, L. (2012). Motives for sharing in social networks. Journal of Development Economics, 99(1):13–26.
- Liu, Y., Chang, C.-C. H., Marsh, G. M., and Wu, F. (2012). Population attributable risk of aflatoxin-related liver cancer: Systematic review and meta-analysis. *European Journal of Cancer*, 48(14):2125–2136.
- Liu, Y. and Wu, F. (2010). Global burden of aflatoxin-induced hepatocellular carcinoma: A risk assessment. *Environmental Health Perspectives*, 118(6):818–824.
- Macchiavello, R. and Miquel-Florensa, J. (2019). Buyer-driven upgrading in GVCs: The Sustainable Quality Program in Colombia. Working Paper.
- Maertens, M. and Swinnen, J. F. (2009). Trade, standards, and poverty: Evidence from Senegal. *World Development*, 37(1):161–178.
- Magnan, N., Hoffmann, V., Gajate Garrido, G., Kanyam, D. A., and Opoku, N. (2021). Information, technology, and market rewards: Incentivizing aflatoxin control in Ghana. *Journal of Development Economics*, page 102620.
- Magruder, J. R. (2018). An assessment of experimental evidence on agricultural technology adoption in developing countries. *Annual Review of Resource Economics*, 10(1):299–316.
- Masters, W. A. (2007). Distortions to agricultural incentives in Senegal. Working Paper 41, World Bank.
- Mbaye, A. A. (2005). Sanitary and phytosanitary requirements and developing-country agro-food exports. Discussion Paper 47846, The World Bank.
- Michler, J. D., Tjernström, E., Verkaart, S., and Mausch, K. (2019). Money matters: The role of yields and profits in agricultural technology adoption. *American Journal of Agricultural Economics*, 101(3):710–731.
- Michler, J. D. and Wu, S. Y. (2020). Relational contracts in agriculture: Theory and evidence. Annual Review of Resource Economics, 12(1):annurev-resource-101719-034514.
- Miguel, E. and Kremer, M. (2004). Worms: Identifying impacts on education and health in the presence of treatment externalities. *Econometrica*, 72(1):159–217.
- Minot, N. and Sawyer, B. (2016). Contract farming in developing countries: Theory, practice, and policy implications. In Devaux, André, Torero, Máximo, Donovan, Jason, and Horton, Douglas, editors, *Innovation for Inclusive Value-Chain Development: Successes and Challenges*. International Food Policy Research Institute.
- Minten, B., Randrianarison, L., and Swinnen, J. F. (2009). Global retail chains and poor farmers: Evidence from Madagascar. *World Development*, 37(11):1728–1741.
- Mutz, D. C., Pemantle, R., and Pham, P. (2019). The perils of balance testing in experimental design: Messy analyses of clean data. *The American Statistician*, 73(1):32–42.

- Nakano, Y. and Magezi, E. F. (2020). The impact of microcredit on agricultural technology adoption and productivity: Evidence from randomized control trial in Tanzania. World Development, 133:104997.
- Narayan, T., Mainville, D., Geyer, J., Hausdorff, K., and Cooley, D. (2019). AgResults impact evaluation report: Nigeria aflasafe challenge project. Technical Report, Abt Associates.
- National Toxicology Program (2016). Report on carcinogens, fourteenth edition. Technical report, US. Department of Health and Human Services, Public Health Service.
- Nourani, V. (2019). Multi-object Social Learning and Technology Adoption in Ghana: Learning from Friends and Reacting to Acquaintances. Working Paper.
- Otsuki, T., Wilson, J. S., and Sewadeh, M. (2001). What price precaution? European harmonisation of aflatoxin regulations and African groundnut exports. *European Review of Agricultural Economics*, 28(2):22.
- Park, S., Yuan, Z., and Zhang, H. (2021). Technology Training, Buyer-Supplier Linkage, and Quality Upgrading in an Agricultural Supply Chain. Working Paper.
- Partnership for Aflatoxin Control in Africa (2015). Aflatoxin impacts and potential solutions in agriculture, trade, and health. Technical Brief.
- Pretari, A., Hoffmann, V., and Tian, L. (2019). Post-harvest practices for aflatoxin control: Evidence from Kenya. Journal of Stored Products Research, 82:31–39.
- Prieto, S., Ricker-Gilbert, J., Bauchet, J., and Sall, M. (2019). Incomplete information and product quality in rural markets: Evidence from an experimental auction for maize in Senegal. *Economic Development and Cultural Change*, Forthcoming:706816.
- Probst, C., Schulthess, F., and Cotty, P. J. (2010). Impact of Aspergillus section Flavi community structure on the development of lethal levels of aflatoxins in Kenyan maize (Zea mays). *Journal of Applied Microbiology*, 108(2):600–610.
- Reardon, T., Barrett, C. B., Berdegué, J. A., and Swinnen, J. F. M. (2009). Agrifood industry transformation and small farmers in developing countries. World Development, 37(11):1717–1727.
- Rigdon, M. (2009). Trust and reciprocity in incentive contracting. Journal of Economic Behavior & Organization, 70(1):93–105.
- Saenger, C., Torero, M., and Qaim, M. (2014). Impact of third-party contract enforcement in agricultural markets-a field experiment in Vietnam. American Journal of Agricultural Economics, 96(4):1220–1238.
- Senghor, L. A., Ortega-Beltran, A., Atehnkeng, J., Callicott, K. A., Cotty, P. J., and Bandyopadhyay, R. (2020). The atoxigenic biocontrol product Aflasafe SN01 is a valuable tool to mitigate aflatoxin contamination of both maize and groundnut cultivated in Senegal. *Plant Disease*, 104(2):510–520.

- Sharma, K. K., Pothana, A., Prasad, K., Shah, D., Kaur, J., Bhatnagar, D., Chen, Z.-Y., Raruang, Y., Cary, J. W., Rajasekaran, K., Sudini, H. K., and Bhatnagar-Mathur, P. (2018). Peanuts that keep aflatoxin at bay: A threshold that matters. *Plant Biotechnology Journal*, 16(5):1024–1033.
- Sirma, A., Lindahl, J., Makita, K., Senerwa, D., Mtimet, N., Kang'ethe, E., and Grace, D. (2018). The impacts of aflatoxin standards on health and nutrition in sub-Saharan Africa: The case of Kenya. *Global Food Security*, 18:57–61.
- Snyder, C. and Zhuo, R. (2020). Sniff tests as a screen in the publication process: Throwing out the wheat with the chaff. Working Paper w25058, National Bureau of Economic Research, Cambridge, MA.
- Sobel, J. (2005). Interdependent preferences and reciprocity. *Journal of Economic Literature*, 43(2):392–436.
- Sun, D., Liu, Y., Grant, J., Long, Y., Wang, X., and Xie, C. (2021). Impact of food safety regulations on agricultural trade: Evidence from China's import refusal data. *Food Policy*, 105:102185.
- Suri, T. (2011). Selection and Comparative Advantage in Technology Adoption. *Econometrica*, 79(1):159–209.
- Suri, T. and Udry, C. (2022). Agricultural Technology in Africa. Journal of Economic Perspectives, 36(1):33–56.
- Tarozzi, A., Desai, J., and Johnson, K. (2015). The impacts of microcredit: Evidence from Ethiopia. American Economic Journal: Applied Economics, 7(1):54–89.
- Tarozzi, A., Mahajan, A., Blackburn, B., Kopf, D., Krishnan, L., and Yoong, J. (2014). Microloans, insecticide-treated bednets, and malaria: Evidence from a randomized controlled trial in Orissa, India. *American Economic Review*, 104(7):1909–1941.
- Treurniet, M. (2021). The Impact of Being Surveyed on the Adoption of Agricultural Technology. *Economic Development and Cultural Change*.
- Turner, P. C., Moore, S. E., Hall, A. J., Prentice, A. M., and Wild, C. P. (2003). Modification of immune function through exposure to dietary aflatoxin in Gambian children. *Environmental Health Perspectives*, 111(2):217–220.
- Turner, P. C., Sylla, A., Gong, Y. Y., Diallo, M. S., Sutcliffe, A. E., Hall, A. J., and Wild, C. P. (2005). Reduction in exposure to carcinogenic aflatoxins by postharvest intervention measures in west Africa: A community-based intervention study. *Lancet* (London, England), 365(9475):1950–1956.
- Udomkun, P., Wiredu, A. N., Nagle, M., Bandyopadhyay, R., Müller, J., and Vanlauwe, B. (2017). Mycotoxins in sub-Saharan Africa: Present situation, socio-economic impact, awareness, and outlook. *Food Control*, 72:110–122.
- USDA Foreign Agricultural Service (2020). World Agricultural Production. Report WAP 9-20.

- Voth-Gaeddert, L. E., Stoker, M., Torres, O., and Oerther, D. B. (2018). Association of aflatoxin exposure and height-for-age among young children in Guatemala. *International Journal of Environmental Health Research*, 28(3):280–292.
- Waliyar, F., Ba, A., Hassan, H., Bonkoungou, S., and Bosc, J. (1994). Sources of resistance to Aspergillus flavus and aflatoxin contamination in groundnut genotypes in West Africa. *Plant Disease*, 78(7):704–708.
- Waliyar, F., Umeh, V., Traore, A., Osiru, M., Ntare, B., Diarra, B., Kodio, O., Vijay Krishna Kumar, K., and Sudini, H. (2015). Prevalence and distribution of aflatoxin contamination in groundnut (Arachis hypogaea L.) in Mali, West Africa. *Crop Protection*, 70:1–7.
- Watson, S., Diedhiou, P., Atehnkeng, J., Dem, A., Bandyopadhyay, R., Srey, C., Routledge, M., and Gong, Y. (2015). Seasonal and geographical differences in aflatoxin exposures in Senegal. World Mycotoxin Journal, 8(4):525–531.
- Watson, S., Moore, S. E., Darboe, M. K., Chen, G., Tu, Y.-K., Huang, Y.-T., Eriksen, K. G., Bernstein, R. M., Prentice, A. M., Wild, C. P., Xu, Y., Routledge, M. N., and Gong, Y. Y. (2018). Impaired growth in rural Gambian infants exposed to aflatoxin: A prospective cohort study. *BMC Public Health*, 18(1):1247.
- Wild, C. (2002). The toxicology of aflatoxins as a basis for public health decisions. *Mutagenesis*, 17(6):471–481.
- Williams, J. H., Phillips, T. D., Jolly, P. E., Stiles, J. K., Jolly, C. M., and Aggarwal, D. (2004). Human aflatoxicosis in developing countries: A review of toxicology, exposure, potential health consequences, and interventions. *The American Journal of Clinical Nutrition*, 80(5):1106–1122.
- World Bank (2015). Etude diagnostique de la chaîne de valeurs arachide au Sénégal. Technical Report ACS16609, World Bank.
- World Bank (2017). Groundnut value chain competitiveness and prospects for development. Technical Report, World Bank, Washington, DC.
- World Bank, editor (2020). World Development Report 2020: Trading for Development in the Age of Global Value Chains. World Bank Group, Washington.

## A Additional detail on design and experimental context

#### A.1 Pre-Specified Analysis

In the interest of transparency, we describe here several ways in which our analysis deviates from the pre-specified analysis, and why. We feel these changes are justified given the deviations in design and implementation from the PAP, but readers are invited to draw their own conclusions and judge our results accordingly.

First, the PAP covered a planned larger trial, featuring multiple treatment groups with both partial and full contracts. After writing this PAP and preparing to launch the project in the field, we decided to simplify the design to include a single treatment group covering the "full contract" described in the PAP. We reasoned this would allow us to demonstrate an upper bound on the treatment effects we might expect from partial contracts, as well as establishing working relationships with our implementation partners. We planned to implement the full design discussed in the PAP in the second year of the project, but this second phase was delayed due to COVID-19.

Second, the PAP hinged on an individually-randomized trial. Upon finalizing our partnerships with two groundnut cooperatives, we learned more about the village-centered way they organize their existing field activities, including distribution of seeds and other inputs on credit. We decided that following this model in our project would significantly facilitate project implementation. More importantly, following this model allowed us to test an intervention which is feasible for our partner cooperatives to implement themselves within their existing model. Additionally, a cluster-randomized trial minimizes the potential for spillovers because both aflatoxin contamination and any potential effects of Aflasafe can impact neighboring fields.

Finally, we modify our empirical strategy given that the trial as implemented included only one treatment and was randomized at the village level. In particular, we are unable to implement our planned strategy for low variation in aflatoxin levels among non-treated farmers, which was to exclude villages with sufficiently low levels of aflatoxin among control farmers. Instead, we test for spatial heterogeneity by interacting treatment status with spatial dummies at cooperative and commune levels. We also pre-specified a number of behavioral mechanisms we would test. Some of these hypotheses are redundant given the simplified treatment. Some are also fruitless to test because our baseline measurements were unable to capture sufficient variation. We discuss some of the issues with baseline measurement above in Section 4.

## **B** Additional tables and figures



Figure B.1: Study regions and sampled villages



Figure B.2: Distribution of test results from IITA agronomic trials in 2019

Note: this figure shows aflatoxin test results from agronomic trials conducted elsewhere in Senegal during the same season we study. Demonstration plots were treated with Aflasafe, whereas comparison plots were not treated.

## B.1 More on baseline and endline data



Figure B.3: Aflasafe adoption





Figure B.4: Output allocation





Figure B.5: Output allocation to the cooperative (kgs)



(a) Northern cooperative

Figure B.6: Post-harvest aflatoxin levels

# B.2 Additional results tables

	(1)	(2)	(3)	(4)
Panel A: ITT estimates (ad	min data adopt	ion)		
Treated	0.79***	$0.75^{***}$	0.74***	0.83***
	(0.060)	(0.070)	(0.090)	(0.060)
Treated x		0.13*		
Beciprocal		(0.070)		
		(0.010)	0.00	
Treated ×			0.06	
Patient			(0.070)	
Treated $\times$				-0.16*
Risk-loving				(0.090)
Reciprocal	0.06**	-0.01	$0.05^{*}$	0.06**
-	(0.030)	(0.040)	(0.030)	(0.030)
Patient	-0.03	-0.03	-0.06	-0.02
1 0010110	(0.030)	(0.030)	(0.050)	(0.030)
	0.00*	0.00*	0.00*	0.15**
Kisk-loving	$(0.00^{+})$	$(0.00^{+})$	$(0.00^{+})$	(0.060)
	(0.030)	(0.030)	(0.030)	(0.060)
Ν	396	396	396	396
Control mean	0.102	0.102	0.102	0.102
Panel B: ITT estimates (sel	f-reported adop	tion)		
Treated	$0.60^{***}$	$0.57^{***}$	$0.59^{***}$	$0.64^{***}$
	(0.060)	(0.070)	(0.100)	(0.070)
Treated $\times$		0.12		
Reciprocal		(0.080)		
Troated V		× /	0.01	
Patient			(0.001)	
			(0.090)	
Treated ×				-0.15
Kısk-loving				(0.110)
Reciprocal	0.01	-0.06	0.01	0.01
	(0.040)	(0.060)	(0.040)	(0.040)
Patient	-0.05	-0.05	-0.06	-0.04
	(0.040)	(0.040)	(0.070)	(0.040)
Risk loving	0.04	0.04	0.04	0 19*
TUSA-IOVIIIg	0.04 (0.060)	(0.04)	0.04 (0.00 0)	(0.12)
	(0.000)	(0.000)	(0.000)	(0.000)
N	370	370	370	370
Control mean	0.197	0.197	0.197	0.197

Table B.1: Adoption of Aflasafe, with behavioral heterogeneity

This table shows results of regressions where the outcome variable is a dummy equal to one if the farmer adopted Aflasafe according to admin and self-reported measures. Panel A presents Intention-To-Treat results where the outcome is regressed on treatment status. Panel B presents 2SLS results where adoption (as measured in cooperative administrative data) is instrumented by treatment status. Column (2) additionally includes an interaction with a dummy equal to one if self-assessed intrinsic reciprocity is greater or equal to 0.6 on a standardized [0,1] scale, column (3) includes an interaction with a dummy equal to one for self-assessed patient farmers, and column (4) includes an interaction with a dummy equal to one if the farmer identified as extremely willing to take risks. In panel B, the 2SLS regressions additionally include treatment interacted with the cooperative dummy as a second instrument. All regressions include commune fixed effects to account for stratified randomization of treatment assignment. Standard errors are clustered at the treatment assignment (village) level.

		Baseline		Pre-Distribution			
	(1)	(2)	(3)	(4)	(5)	(6)	
Treated	0.07***	0.07***	0.06***	0.20***	0.22***	0.20***	
	(0.02)	(0.02)	(0.02)	(0.05)	(0.04)	(0.05)	
Observations	396	396	396	396	396	396	
$R^2$	0.024	0.090	0.198	0.060	0.082	0.124	
Control Mean Dep. Var	0.91	0.91	0.91	0.68	0.68	0.68	
Commune FE	Ν	Υ	Υ	Ν	Υ	Υ	
Baseline controls	Ν	Ν	Υ	Ν	Ν	Υ	

Table B.2: Aflasafe non-binding intent to adopt

Results in this table are from linear regressions of the intention to adopt dummies on the treatment dummy. Baseline intent to adopt was elicited immediately after farmers were informed about Aflasafe and their treatment assignment. Pre-Distribution intent to adopt was elicited by phone when Aflasafe distribution began. Standard errors (in parentheses) are clustered at the treatment assignment (village cluster) level. Baseline controls included are all variables shown above in Table 1.

	(1)	(2)	(3)	(4)	(5)	(6)
Panal A. ITT actimates	(-)	(-)	(0)	(-)	(0)	(0)
Treated	0.08	0.08	-0.03	-0.03	-0.08	-0.06
1100000	(0.050)	(0.050)	(0.060)	(0.070)	(0.070)	(0.070)
Treated x	· · · ·	. ,	0.33***	0.32**	· /	
Pred. high risk (LASSO)			(0.110)	(0.140)		
Treated x				· · · ·	0.25**	0.24
Pred. high risk (quadratic)					(0.130)	(0.160)
Pred high risk (LASSO)			-0.34***	-0.19	. ,	. ,
			(0.090)	(0.120)		
Pred. high risk (quadratic)					-0.39***	-0.32***
Troat ingh tion (quadratic)					(0.080)	(0.110)
Baseline controls	Ν	Y	Ν	Y	N	Y
Ν	328	328	328	328	328	328
Panel B: TOT estimates (a	dmin data	adoption	)			
Adopted	0.10	0.10*	-0.05	-0.04	-0.11	-0.09
	(0.070)	(0.060)	(0.080)	(0.090)	(1.260)	(0.180)
Adopted $\times$			$0.38^{***}$	$0.38^{***}$		
Pred. high risk (LASSO)			(0.130)	(0.140)		
Adopted $\times$					0.33	0.31
Pred. high risk (quadratic)					(1.280)	(0.240)
Pred. high risk (LASSO)			-0.34***	-0.19		
			(0.090)	(0.120)		
Pred. high risk (quadratic)					-0.39***	-0.32***
					(0.080)	(0.110)
Baseline controls	Ν	Υ	Ν	Υ	Ν	Υ
N	328	328	328	328	328	328
Panel C: TOT estimates (se	elf-reported	d adoption	<i>v)</i>			
Adopted	0.14	$0.14^{*}$	-0.11	-0.14	-0.10	-0.09
	(0.080)	(0.070)	(0.150)	(0.980)	(0.100)	(0.100)
Adopted $\times$			$0.72^{*}$	0.84		
Pred. nign risk (LASSO)			(0.390)	(3.000)		
Adopted ×					0.39	0.40
Pred. nign risk (quadratic)					(0.280)	(0.310)
Pred. high risk (LASSO)			$-0.34^{***}$	-0.19		
			(0.090)	(0.120)		
Pred. high risk (quadratic)					$-0.39^{***}$	$-0.32^{***}$
		<b>.</b>		<b>.</b>	(0.080)	(0.110)
Baseline controls	N 207	Y 207	N 207	Y 207	N 207	Y 207
Control mean	0.659	0.659	0.659	0.659	0.659	0.659

 Table B.3: Phytosanitary standard compliance

This table shows results of regressions where the outcome variable is a dummy equal to one if the groundnut sample complied with EU phytosanitary standards. Panel A presents Intention-To-Treat results where the outcome is regressed on treatment status. Panel B presents 2SLS results where adoption (as measured in cooperative administrative data) is instrumented by treatment status. Panel C presents 2SLS results where adoption (as reported in the endline survey) is instrumented by treatment status. Columns (3) and (4) additionally include an interaction with the cooperative dummy. Columns (5) and (6) include an interaction with a dummy equal to one if the village was predicted to be at high risk given agro-climatic conditions experienced during the growing season. See Appendix C for details. In panels B and C, the 2SLS regressions additionally include treatment interacted with the cooperative dummy or the predicted high risk dummy as a second instrument. Standard errors are clustered at the treatment assignment (village) level and bootstrapped in columns (3)-(6). Baseline controls included are all variables shown above in Table 1

	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: ITT estimates	0.01	0.10		0.40		
Treated	-0.21 (0.200)	-0.18 (0.170)	(0.22) (0.240)	(0.19)	(0.31) (0.230)	(0.27) (0.330)
Treated $\times$	()	()	-1.24***	-1.21***	()	()
Pred. high risk (LASSO)			(0.370)	(0.460)		
Treated $\times$					-0.87*	-0.82
Pred. high risk (quadratic)			1 10444	1 1044	(0.500)	(0.550)
Pred. high risk (LASSO)			(0.290)	(0.430)		
Pred. high risk (quadratic)			. ,	. ,	1.22***	0.88**
· · ·					(0.380)	(0.430)
Baseline controls	N	Y	N	Y 2200	N	Y
N <u>     Remail Di TOT estimates (a</u>	328	328	328	328	328	328
Adopted	-0.27	-0.23	0.29	0.25	0.43	0.38
	(0.240)	(0.200)	(0.280)	(0.310)	(0.310)	(0.440)
Adopted $\times$			-1.44***	-1.41***		
Pred. high risk (LASSO)			(0.450)	(0.540)		
Adopted × Pred high risk (quadratic)					$-1.14^{*}$	-1.09 (0.780)
Pred high risk (LASSO)			1 40***	1 10**	(0.000)	(0.100)
			(0.290)	(0.430)		
Pred. high risk (quadratic)					$1.22^{***}$	0.88**
					(0.380)	(0.430)
Baseline controls	N	Y	N	Y 200	N	Y
Panal C: TOT actimates (a)	J20	320 Ladontion	)	320	328	328
Adopted	-0.34	-0.30	0.30	0.29	0.57	0.57
-	(0.330)	(0.260)	(0.360)	(0.310)	(0.450)	(0.370)
Adopted $\times$			-1.82**	-1.87**		
Pred. high risk (LASSO)			(0.820)	(0.880)		
Adopted $\times$ Pred. high risk (quadratic)					-1.67 (1.020)	-1.71 (1.190)
Pred. high risk (LASSO)			$1.40^{***}$ (0.290)	$1.10^{**}$ (0.430)		
Pred. high risk (quadratic)			. /	. ,	1.22***	0.88**
/					(0.380)	(0.430)
Baseline controls	N	Y	N	Y	N	Y
N Control mean	$\frac{307}{1.304}$	$\frac{307}{1.304}$	$\frac{307}{1.304}$	$\frac{307}{1.304}$	$\frac{307}{1.304}$	$\frac{307}{1.304}$

Table B.4: Natural log of measured aflatoxin contamination levels

This table shows results of regressions where the outcome variable is the natural log of the measured aflatoxin contamination (in parts per billion) in the tested groundnut sample. Panel A presents Intention-To-Treat results where the outcome is regressed on treatment status. Panel B presents 2SLS results where adoption (as measured in cooperative administrative data) is instrumented by treatment status. Panel C presents 2SLS results where adoption (as reported in the endline survey) is instrumented by treatment status. Columns (3) and (4) additionally include an interaction with the cooperative dummy. Columns (5) and (6) include an interaction with a dummy equal to one if the village was predicted to be at high risk given agro-climatic conditions experienced during the growing season. See Appendix C for details. In panels B and C, the 2SLS regressions additionally include treatment interacted with the cooperative dummy or the predicted high risk dummy as a second instrument. Standard errors are clustered at the treatment assignment (village) level and bootstrapped in columns (3)-(6). Baseline controls include are all variables shown above in Table 1

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Panel A: ITT estimates Treated	$0.12^{**}$ (0.050)	$0.09^{*}$ (0.050)	0.04 (0.050)	0.01 (0.050)	0.00 (0.080)	-0.05 (0.090)	$0.15^{**}$ (0.060)	$0.12^{**}$ (0.050)
Treated $\times$ Reciprocal			$0.27^{***}$ (0.090)	$0.27^{***}$ (0.090)				
Treated $\times$ Patient					$0.16^{*}$ (0.090)	$0.18^{*}$ (0.100)		
Treated $\times$ Risk loving							-0.10 (0.090)	-0.10 (0.090)
Reciprocal	-0.02 (0.050)	0.01 (0.060)	$-0.17^{***}$ (0.050)	$-0.13^{**}$ (0.050)	-0.03 (0.050)	$\begin{array}{c} 0.01 \\ (0.060) \end{array}$	-0.02 (0.050)	0.01 (0.060)
Patient	0.00 (0.050)	-0.03 (0.060)	-0.01 (0.050)	-0.04 (0.060)	-0.08 (0.070)	-0.12 (0.080)	$0.00 \\ (0.050)$	-0.03 (0.060)
Risk loving	$0.11^{**}$ (0.050)	$0.11^{*}$ (0.060)	$0.11^{**}$ (0.050)	$0.10^{*}$ (0.060)	$0.11^{*}$ (0.050)	$0.10^{*}$ (0.060)	$0.17^{**}$ (0.070)	$0.16^{**}$ (0.070)
Panel B: TOT estimates (adm Adopted	$ \begin{array}{c} in \ data \ ado \\ 0.15^{***} \\ (0.060) \end{array} $	$ption) \\ 0.12^{**} \\ (0.050)$	$0.05 \\ (0.070)$	0.01 (0.070)	-0.01 (0.110)	-0.07 (0.110)	$0.18^{***}$ (0.060)	$0.14^{**}$ (0.060)
Adopted $\times$ Reciprocal			$0.31^{***}$ (0.110)	$\begin{array}{c} 0.31^{***} \\ (0.110) \end{array}$				
Adopted $\times$ Patient					$0.21^{*}$ (0.120)	$0.23^{*}$ (0.120)		
Adopted $\times$ Risk loving							-0.12 (0.130)	-0.11 (0.120)
Reciprocal	-0.03 (0.050)	0.01 (0.050)	$-0.21^{***}$ (0.070)	$-0.17^{**}$ (0.070)	-0.04 (0.050)	$0.00 \\ (0.050)$	-0.03 (0.050)	0.01 (0.050)
Patient	0.00 (0.050)	-0.03 (0.050)	-0.01 (0.050)	-0.03 (0.050)	-0.11 (0.080)	$-0.15^{*}$ (0.090)	$0.01 \\ (0.050)$	-0.02 (0.050)
Risk loving	$0.10^{*}$ (0.050)	$0.10^{*}$ (0.050)	$0.11^{**}$ (0.050)	$0.10^{*}$ (0.050)	$0.10^{*}$ (0.050)	$0.09 \\ (0.060)$	$0.17^{**}$ (0.090)	$0.16^{*}$ (0.080)
Panel C: TOT estimates (self- Adopted	reported add $0.20^{**}$ (0.080)	$\begin{array}{c} (0.070) \\ 0.15^{**} \\ (0.070) \end{array}$	0.06 (0.090)	0.01 (0.090)	-0.02 (0.140)	-0.09 (0.150)	$0.23^{***}$ (0.090)	$0.18^{**}$ (0.080)
Adopted $\times$ Reciprocal			$0.39^{***}$ (0.140)	$0.38^{***}$ (0.140)				
Adopted $\times$ Patient					$0.28^{*}$ (0.150)	$0.30^{*}$ (0.160)		
Adopted $\times$ Risk loving							-0.15 (0.180)	-0.15 (0.170)
Reciprocal	-0.02 (0.050)	0.01 (0.050)	$-0.23^{***}$ (0.080)	$-0.19^{**}$ (0.080)	-0.03 (0.050)	$0.00 \\ (0.050)$	-0.02 (0.050)	0.01 (0.050)
Patient	0.01 (0.050)	-0.02 (0.060)	-0.01 (0.050)	-0.04 (0.050)	-0.14 (0.100)	$-0.18^{*}$ (0.100)	$0.01 \\ (0.050)$	-0.02 (0.050)
Risk loving	$0.11^{**}$ (0.050)	$0.10^{*}$ (0.060)	$0.10^{*}$ (0.050)	0.09 (0.060)	$0.11^{**}$ (0.050)	$0.10^{*}$ (0.060)	$0.19^{*}$ (0.110)	$0.18^{*}$ (0.100)
Baseline controls N Control mean	N 370 0.230	Y 370 0.230	N 370 0.230	Y 370 0.230	N 370 0.230	Y 370 0.230	N 370 0.230	Y 370 0.230

#### Table B.5: Sold any output to coop, with behavioral heterogeneity

This table shows results of regressions where the outcome variable is a dummy equal to one if the farmer reported selling any output to the cooperative. Panel A presents Intention-To-Treat results where the outcome is regressed on treatment status. Panel B presents 2SLS results where adoption (as measured in cooperative administrative data) is instrumented by treatment status. Panel C presents 2SLS results where adoption (as reported in the endline survey) is instrumented by treatment status. Columns (3) and (4) additionally include an interaction with a dummy equal to one if self-assessed intrinsic reciprocity is greater or equal to 0.6 on a standardized [0,1] scale, columns (5) and (6) include an interaction with a dummy equal to one if the farmer identified as extremely willing to take risks. In panels B and C, the 2SLS regressions additionally include treatment interacted with the cooperative dummy as a second instrument. Standard errors are clustered at the treatment assignment (village) level. Baseline controls include are all variables shown above in Table 1

	(1)	(2)	(2)	(4)	(E)	(6)	(7)	(9)
	(1)	(2)	(3)	(4)	(5)	(0)	(7)	(8)
Panel A: ITT estimates Treated	$0.85^{**}$ (0.340)	$0.64^{*}$ (0.320)	0.27 (0.360)	0.08 (0.370)	0.04 (0.600)	-0.27 (0.620)	$1.04^{**}$ (0.390)	$0.81^{**}$ (0.380)
Treated $\times$ Reciprocal			$1.86^{***}$ (0.600)	$\begin{array}{c} 1.79^{***} \\ (0.610) \end{array}$				
Treated $\times$ Patient					1.02 (0.650)	1.14 (0.690)		
Treated $\times$ Risk-loving							-0.73 (0.650)	-0.69 (0.620)
Reciprocal	-0.14 (0.370)	0.11 (0.410)	$-1.13^{***}$ (0.350)	$-0.85^{**}$ (0.380)	-0.16 (0.370)	0.08 (0.410)	-0.14 (0.370)	0.09 (0.410)
Patient	-0.04 (0.350)	-0.26 (0.390)	-0.10 (0.350)	-0.31 (0.380)	-0.56 (0.490)	-0.83 (0.530)	$\begin{array}{c} 0.00 \\ (0.340) \end{array}$	-0.23 (0.380)
Risk-loving	$0.83^{**}$ (0.380)	$0.76^{*}$ (0.400)	$0.83^{**}$ (0.370)	$0.74^{*}$ (0.390)	$0.80^{**}$ (0.390)	$0.72^{*}$ (0.400)	$1.24^{**}$ (0.480)	$1.15^{**}$ (0.490)
Panel B: TOT estimates (adm	in data ado	ption)						
Adopted	$1.08^{***}$ (0.400)	$0.81^{**}$ (0.380)	0.34 (0.470)	0.09 (0.470)	$\begin{array}{c} 0.01 \\ (0.770) \end{array}$	-0.42 (0.790)	$\begin{array}{c} 1.26^{***} \\ (0.450) \end{array}$	$0.99^{**}$ (0.430)
Adopted $\times$ Reciprocal			$2.11^{***}$ (0.740)	$2.06^{***}$ (0.730)				
Adopted $\times$ Patient					$1.32 \\ (0.810)$	$1.50^{*}$ (0.860)		
Adopted $\times$ Risk-loving							-0.84 (0.890)	-0.82 (0.800)
Reciprocal	-0.19 (0.360)	$0.06 \\ (0.380)$	$-1.39^{***}$ (0.470)	$-1.10^{**}$ (0.480)	-0.23 (0.360)	$\begin{array}{c} 0.03 \\ (0.380) \end{array}$	-0.20 (0.350)	$\begin{array}{c} 0.06 \\ (0.380) \end{array}$
Patient	0.01 (0.340)	-0.22 (0.370)	-0.08 (0.320)	-0.28 (0.360)	-0.71 (0.560)	$-1.01^{*}$ (0.590)	$\begin{array}{c} 0.03 \ (0.330) \end{array}$	-0.20 (0.360)
Risk-loving	$0.75^{**}$ (0.380)	$0.70^{*}$ (0.380)	$0.77^{**}$ (0.370)	$0.71^{*}$ (0.380)	$0.73^{*}$ (0.390)	$0.66^{*}$ (0.390)	$1.25^{**}$ (0.610)	$1.18^{**}$ (0.580)
Panel C: TOT estimates (self-	reported add	option)						
Adopted	$1.41^{**}$ (0.550)	$1.05^{**}$ (0.500)	0.45 (0.640)	0.10 (0.620)	-0.02 (1.010)	-0.52 (1.070)	$\begin{array}{c} 1.62^{***} \\ (0.590) \end{array}$	$1.26^{**}$ (0.540)
Adopted $\times$ Reciprocal			$2.65^{***}$ (0.930)	$2.57^{***}$ (0.920)				
Adopted $\times$ Patient					1.78 (1.090)	$1.93^{*}$ (1.150)		
Adopted $\times$ Risk-loving							-1.05 (1.280)	-1.07 (1.140)
Reciprocal	-0.15 (0.360)	0.07 (0.380)	$-1.54^{***}$ (0.580)	$-1.26^{**}$ (0.560)	-0.20 (0.360)	$\begin{array}{c} 0.02 \\ (0.380) \end{array}$	-0.12 (0.360)	$\begin{array}{c} 0.11 \\ (0.380) \end{array}$
Patient	$0.03 \\ (0.350)$	-0.20 (0.380)	-0.08 (0.340)	-0.30 (0.370)	-0.92 (0.660)	$-1.21^{*}$ (0.710)	$0.04 \\ (0.340)$	-0.19 (0.370)
Risk-loving	$0.78^{**}$ (0.380)	$0.72^{*}$ (0.380)	$0.72^{*}$ (0.380)	$0.66^{*}$ (0.390)	$0.77^{**}$ (0.380)	$0.70^{*}$ (0.390)	$1.36^{*}$ (0.770)	$1.31^{*}$ (0.700)
Baseline controls	N	Y	Ν	Y	N	Y	Ν	Y
N Control mean	$370 \\ 1.583$	$370 \\ 1.583$	$370 \\ 1.583$	$370 \\ 1.583$	$370 \\ 1.583$	$370 \\ 1.583$	$370 \\ 1.583$	$370 \\ 1.583$

#### Table B.6: Quantity sold to coop, with behavioral heterogeneity

This table shows results of regressions where the outcome variable is the inverse hyperbolic sine transformation of groundnut output (in kgs) sold to the cooperative.. Panel A presents Intention-To-Treat results where the outcome is regressed on treatment status. Panel B presents 2SLS results where adoption (as measured in cooperative administrative data) is instrumented by treatment status. Panel C presents 2SLS results where adoption (as reported in the endline survey) is instrumented by treatment status. Columns (3) and (4) additionally include an interaction with a continuous measure of reciprocity bounded in [0,1], columns (5) and (6) include an interaction with a dummy equal to one for self-assessed patient farmers, and columns (7) and (8) include an interaction with a dummy equal to one if the farmer identified as extremely willing to take risks. In panels B and C, the 2SLS regressions additionally include treatment interacted with the cooperative dummy as a second instrument. Standard errors are clustered at the treatment assignment (village) level. Baseline controls include are all variables shown above in Table 1

## C Agro-climatic conditions and aflatoxin contamination

	(1)	(2)	(3)
3-day dry spells	1.01 (0.68)		$1.34^{**}$ (0.56)
$(3-\text{day dry spells})^2$	-0.01 (0.01)		$-0.02^{**}$ (0.01)
Max temperature		$-5.09^{**}$ (2.08)	$-6.45^{***}$ (2.11)
$(Max temperature)^2$		$0.06^{**}$ (0.02)	$0.07^{***}$ (0.02)
Observations $R^2$	$173 \\ 0.133$	$173 \\ 0.147$	$173 \\ 0.165$

Table C.1: Agro-climatic predictors of aflatoxin contamination

Results in this table are from linear regressions of the observed aflatoxin level (log-transformed) among control farmers on agro-climatic variables. 3-day dry spells is the number of 3-day dry spells observed during the growing season. Max temperature is the mean value of the max temperature observed in dekadal observations over the growing season. All regressions include commune fixed effects. Standard errors (in parentheses) are clustered at the village level.

	(1) Dried on ground	(2) Used standard storage
Treated=1	0.00 (.)	$0.01 \\ (0.01)$
Northern cooperative	-0.03 (0.03)	-0.05 (0.03)
Treated=1 $\times$ Northern cooperative	$0.01 \\ (0.01)$	-0.02 (0.02)
$\frac{\text{Observations}}{R^2}$	$353 \\ 0.031$	$353 \\ 0.020$

Table C.2: Post-harvest practices at endline

Results in this table are from linear regressions of self-reported post-harvest practices at endline on the treatment dummy and the cooperative dummy. Dried on ground is a dummy equal to one if the farmer dried his groundnuts directly on the ground, as opposed to a tarp or concrete pad. Used standard storage is a dummy equal to one if the farmer reported using standard single-layer plastic bags to store harvested groundnuts. All regressions include commune fixed effects. Standard errors (in parentheses) are clustered at the village level.

### D Baseline details

#### D.1 Aflasafe script

The following is the script that was presented to farmers in Wolof, translated to English, at the end of the baseline survey:

Now, we would like to talk about aflatoxin and a new product called Aflasafe. Aflatoxin is produced by a fungus that comes from the soil and grows on groundnuts, maize, and other crops. When crops are not dried well, or not stored in dry conditions, this can cause aflatoxin to increase and spread within your stored crops. Aflatoxin has many negative health effects, especially for pregnant women and young children, and can cause liver cancer when consumed in large amounts over time.

Aflasafe is a new product developed to fight aflatoxin. It is a biological product, not chemical, and uses a non-toxigenic fungus to compete against the toxic fungus which produces aflatoxin on crops. It was originally developed by scientists in America and Nigeria, and customized for use in Senegal. It has been tested here for more than five years, and this year is now launching for sale in the market. If you use Aflasafe correctly, it has been shown to reduce aflatoxin levels in crops by 80-100%. It can also help protect your crops during storage.

Aflasafe is not a substitute for correctly drying and storing your groundnuts. But when used together with these good practices, it can make your groundnuts safer to eat. Buyers and exporters are also interested in buying groundnuts without aflatoxin. Aflasafe is designed to be applied in your field, about 6 weeks after planting, just after the last weeding before the flowering. Aflasafe is distributed as a blue coating on sterilized sorghum seeds which will not grow. To apply Aflasafe, you walk around your fields and broadcast the same way as fertilizer but a small amount of seeds evenly. To treat one hectare of groundnuts, you would need to use 10 kg of Aflasafe. The market price of 10 kgs is 10000 CFA.

Here is a video with more information about Aflasafe: video link (Wolof)

In partnership with the cooperative, we are making 10 kgs of Aflasafe available for you to purchase. To purchase the Aflasafe, you would visit the cooperative's magasin to pick it up. The Aflasafe will be available from the magasin before the end of July. You should call to confirm it is available before traveling to pick it up. In addition, we will offer a free service to test your production and certify it if it is low in aflatoxin. We will offer this in your normal collection point with the cooperative.

#### **Treated farmers**

We have a coupon here that allows you to receive 10 kgs of Aflasafe this month, and pay 10000 CFA in kind when you bring your groundnuts to the buying center. You do not decide now if you want the Aflasafe, but you will need to decide and pick it up at the magasin before September 1. The cooperative will distribute Aflasafe only to those who bring their coupon to the supply shop at the collection point, so make sure to bring this with you.

In addition, the cooperative will send an animateur to your field to help answer any questions you may have about how to properly apply Aflasafe, including the correct time to apply it. They will also verify for the cooperative that you applied Aflasafe to your field.

If your groundnuts pass the low-aflatoxin test and receive quality certification, the buyers at the center will pay you a guaranteed bonus of 40 CFA/kg over the price the cooperative normally pays. If you treat your field with Aflasafe, it is important that you keep those treated groundnuts separate from any others. If you mix them together with untreated groundnuts, it will affect the results of the aflatoxin test.

Do you plan to accept this contract and pick up your Aflasafe from the cooperative?

#### **Control farmers**

We have a coupon here that allows you to purchase 10 kgs of Aflasafe from the cooperative supply shop for 10000 CFA. You do not need to decide now if you want the Aflasafe, but you will need to decide and purchase it at the magasin before September 1. The cooperative will distribute Aflasafe only to those who bring their coupon to the supply shop at the collection point, so make sure to bring this with you.

The cooperative may send an animateur to visit your field to verify if you used Aflasafe.

If your groundnuts pass the low-aflatoxin test and receive quality certification, there may be buyers available who will pay more than the normal cooperative price. If you treat your field with Aflasafe, it is important that you keep those treated groundnuts separate from any others. If you mix them together with untreated groundnuts, it will affect the results of the aflatoxin test.

Do you plan to purchase the Aflasafe from the cooperative?

#### D.2 Behavioral variable measurement

#### Reciprocity

We define reciprocity as the mean response to the following three questions, for which the answer options were (Always willing / Sometimes willing / Never willing).

- When someone does me a favor, I am willing to return it
- How willing are you to punish someone who treats you unfairly, even if there may be costs for you?
- How willing are you to punish someone who treats others unfairly, even if there may be costs to you?

#### Patience

We define patience if a respondent responded "Yes, always" or "Yes, sometimes" to the following question:

• In comparison to others, are you a person who is generally willing to give up something today in order to benefit from that in the future?

### Risk

Following Charness and Viceisza (2016), we elicited risk aversion using an 11-point scale:

• Please tell me, in general, how willing or unwilling you are to take risks, using a scale from 0 to 10, where 0 means you are "completely unwilling to take risks" and 10 means you are "very willing to take risks." You can also use any number between 0 and 10 to indicate where you fall on the scale, using 0, 1, 2, 3, 4, 5, 6, 7, 8, 9, or 10